# ESSAYS IN INTERNATIONAL FINANCE

No. 167, June 1987

# AN INCREASING ROLE FOR THE ECU: A CHARACTER IN SEARCH OF A SCRIPT

# RAINER STEFANO MASERA



# INTERNATIONAL FINANCE SECTION

DEPARTMENT OF ECONOMICS PRINCETON UNIVERSITY PRINCETON, NEW JERSEY

# ESSAYS IN INTERNATIONAL FINANCE

ESSAYS IN INTERNATIONAL FINANCE are published by the International Finance Section of the Department of Economics of Princeton University. The Section sponsors this series of publications, but the opinions expressed are those of the authors. The Section welcomes the submission of manuscripts for publication in this and its other series. See the Notice to Contributors at the back of this Essay.

The author of this Essay, Rainer Stefano Masera, is the Central Manager for Economic Research of the Bank of Italy and a member of the Deputies of the Group of Ten and of the Committee of Alternates of the European Community Central Banks. He has taught at the Universities of Bergamo and Rome. In addition to articles in professional journals, he has written several books, the most recent being L'Unificazione Monetaria e lo SME (Monetary Integration and the EMS) (1980).

> PETER B. KENEN, Director International Finance Section

# ESSAYS IN INTERNATIONAL FINANCE No. 167, June 1987

# AN INCREASING ROLE FOR THE ECU: A CHARACTER IN SEARCH OF A SCRIPT

# RAINER STEFANO MASERA



# INTERNATIONAL FINANCE SECTION

DEPARTMENT OF ECONOMICS PRINCETON UNIVERSITY PRINCETON, NEW JERSEY

# INTERNATIONAL FINANCE SECTION EDITORIAL STAFF

Peter B. Kenen, Director Ellen Seiler, Editor Carolyn Kappes, Editorial Aide Barbara Radvany, Subscriptions and Orders

# Library of Congress Cataloging-in-Publication Data

Masera, R. S. An increasing role for the ECU

(Essays in international finance, ISSN 0071-142X ; no. 167) Bibliography: p. 1. European currency unit. I. Title. II. Series HG136.P7 no. 167 332'.042s 87-3747 [HG930.5] [332.4'56094] ISBN 0-88165-074-9

Copyright @ 1987 by International Finance Section, Department of Economics, Princeton University.

All rights reserved. Except for brief quotations embodied in critical articles and reviews, no part of this publication may be reproduced in any form or by any means, including photocopy, without written permission from the publisher.

Printed in the United States of America by Princeton University Press at Princeton, New Jersey.

International Standard Serial Number: 0071-142X International Standard Book Number: 0-88165-074-9 Library of Congress Catalog Card Number: 87-3747

# CONTENTS

•

|   | ACKNOWLEDGMENTS  | v         |
|---|--|-----------|
|   | GLOSSARY   | vi        |
| 1 | INTRODUCTION   | 1         |
| 2 | A BRIEF OVERVIEW OF THE DEVELOPMENT OF THE OFFICIAL ECU        | 2         |
| 3 | THE RAPID DEVELOPMENT OF THE PRIVATE ECU $\stackrel{\prime}{}$ | 6         |
|   | The Private ECU Banking Market                                 | 6         |
|   | The Private ECU Bond Market                                    | 11        |
|   | The Private ECU Exchange Market                                | 11        |
| 4 | REASONS FOR THE GROWTH OF THE PRIVATE-ECU MARKET               | 13        |
| 5 | AN ASSESSMENT OF THE PRESENT AND FUTURE STATUS                 |           |
|   | OF THE ECU   | 17        |
| 6 | THE ECU AS MEANS OF INTERNATIONAL PAYMENTS:                    |           |
|   | A SUGGESTED SCHEME   | 19        |
| 7 | BALANCE-OF-PAYMENTS, MONETARY, AND CREDIT IMPLICATIONS         |           |
|   | OF INCREASED USE OF THE PRIVATE ECU                            | <b>24</b> |
|   | Balance-of-Payments Implications                               | 24        |
|   | Monetary and Credit Implications                               | 26        |
| 8 | CONCLUSIONS  | 28        |
|   | REFERENCES   | 31        |
|   | STATISTICAL APPENDIX: EFFICIENT PORTFOLIO CHOICE               |           |
|   | AND THE ECU  | 34        |

#### ACKNOWLEDGMENTS

This essay is an offshoot of work on the role and prospects of the ECU that is being conducted by the Committee of Alternates of the European Community Central Banks, under mandate of the Governors. The views expressed here are mine, however, and do not represent the official views of the Bank of Italy. I profited from discussions held in the Committee of Alternates under the Chairmanship of André Szasz in 1985 and Anthony Loehnis in 1986. The work of the Alternates was supported by excellent ad hoc reports prepared by the two groups of foreign-exchange and monetary experts, chaired, respectively, by Henning Dalgaard and Robert Raymond. I also benefited from an exchange of views with many colleagues at the Bank of Italy, in particular Emilio Barone, Giannandrea Falchi, Salvatore Rebecchini, and Ambrogio Rinaldi. A preliminary version of this essay was presented at the Conference on the ECU Market: Current Developments and Future Prospects. held at New York University in January 1986. I have greatly benefited from comments by the discussant, Andrea Sommariva. I am also grateful for the comments of an anonymous referee.

|       | GLOSSARY   |
|-------|--|
| BIS   | = Bank for International Settlements                           |
| EC    | = European Community   |
| ECU   | = European Currency Unit                                       |
| EMCF  | = European Monetary Cooperation Fund                           |
| EMS   | = European Monetary System                                     |
| ERM   | = Exchange Rate Mechanism                                      |
| EUA   | = European Unit of Account                                     |
| MESA  | = Mutual ECU Settlement Account                                |
| SWIFT | = Society for Worldwide Interbank Financial Telecommunications |
| VSTF  | = Very-short-term financing                                    |

# AN INCREASING ROLE FOR THE ECU: A CHARACTER IN SEARCH OF A SCRIPT

## **1** Introduction

The European Currency Unit has two aspects—official and private. The link between the two is that the value of both is fixed to the same basket of European Community (EC) currencies. They differ, however, with respect to the way they are created, the credit standing of the issuing entities, the returns they offer, and their usability. The official ECU, which is governed by the rules of the European Monetary System (EMS), is used as an international financial instrument, but only in a restricted sense and only by EC central banks, which hold official-ECU positions vis-à-vis the European Monetary Cooperation Fund (EMCF). The private ECU is basically a Eurocurrency that has acquired the character of an international currency, especially for EC market operators, albeit still on a relatively modest scale. Recently, some EC central banks have also built up holdings of ECU deposits with the international banking system. Although these are clearly part of the foreign-exchange reserves of the central banks, they must still be regarded as "private" ECUs and not "official" ones as previously defined.<sup>1</sup>

This essay focuses on the growing use and future prospects of the ECU. It points out the advantages of establishing a more direct link between the official and the private sides of the ECU so that the ECU can acquire more of the attributes of both a reserve asset and a common European currency, and it explores the ways and means of doing so.

Section 2 briefly reviews the development of the ECU in the official field and section 3 its development in private markets. The advances made so far by the official ECU have fallen short of the original objective of making the ECU "the center of the EMS." Nevertheless, its role has been far from insignificant, and a potentially important package of measures designed to improve its usability has recently been adopted. By contrast, the development of the private ECU in both credit and foreign-exchange markets has exceeded the expectations of most observers.

While a number of factors help explain the rapid growth of the private-ECU market, section 4 argues that the main one is the rational portfolio behavior of European operators engaged in international financial transactions. Standard "efficient portfolio" analysis covering the period of operation of the

<sup>&</sup>lt;sup>1</sup> See facing page for a glossary of acronyms used in this essay.

EMS shows that the ECU has attractive properties, especially for risk-averse investors, in countries with strong currencies as well as in countries with weak ones.

Section 5 assesses the two most important suggestions now under consideration for giving the ECU a central role in the EMS, (1) that the ECU be developed into a full-fledged international reserve asset, and (2) that it gradually acquire the status of a true European currency, functioning in parallel with domestic currencies. My contention is that the two schemes should be viewed as complementary.

International reserve assets must be liquid and fully convertible into other currencies, represent a stable store of value, and provide a competitive return. The portfolio results reported in section 4 suggest that the private ECU has some of these properties while the official ECU has important defects, notably in terms of transferability and convertibility. Following an approach originally suggested by Kenen (1983) for the SDR, section 6 presents a scheme that could be carried out without institutional changes to improve the properties of the ECU by making the official and private ECUs interchangeable.

The counterpart of an expanding use of the official ECU would be its increasing role as a substitute for domestic-currency borrowing and lending, leading to its use as a parallel common currency. Section 7 discusses problems of money and credit control that might arise and points out major differences between the Euromarkets and the ECU.

Section 8 concludes that the proposed two-pronged development of the ECU could become an important instrument with which to foster European monetary integration without threatening monetary stability and credit control. It stresses, however, that this would be so if, and only if, EC governments and citizens did not see such advances as *alternatives* to the necessary adjustment of domestic monetary conditions.

## **2** A Brief Overview of the Development of the Official ECU

According to the December 5, 1978, Brussels resolution of the European Council that established the EMS, its purpose was the "creation of closer monetary cooperation leading to a zone of monetary stability in Europe." The resolution also stated that the ECU would be at the center of the EMS. It would be used as the numeraire for the exchange-rate mechanism, as the basis for detecting divergencies between Community currencies, as the unit of account for the operations of intervention and credit mechanisms, and as a means of settlement between monetary authorities of the European Community. To serve the last function, a stock of "primary" ECUs would be created through revolving swap arrangements. Participating central banks would obtain ECUs by depositing 20 percent of their gold and dollar reserves with the EMCF (Committee of Governors, 1985, pp. 13 ff.).

The 1978 Brussels resolution was a compromise between two models of the EMS that had been explored in the technical discussions preceding the establishment of the system. The first model would have used the ECU as the parameter for defining central rates and intervention obligations around prescribed margins. Under this scheme, only one currency at a time would have reached the prescribed margin. The second model, based on the workings of the earlier exchange-rate mechanism known as the "snake," relied on bilateral central and intervention rates. The second approach was adopted, but as a compromise a "divergence indicator" was introduced to single out the deviating currency upon which the burden of adjustment would primarily fall (EC Monetary Committee, 1978, and Ludlow, 1982, Chap. 6.2).

In spite of the considerable technical ingenuity that went into its construction, the divergence indicator gradually lost importance, in part as a result of increased reliance on discretionary unilateral intervention before exchange rates reached the margins for compulsory intervention. The system has therefore evolved *de facto* in the direction of the second model, but there has been a decline in the importance of obligatory interventions at the points defined by the margins around the bilateral central exchange rates. From a formal point of view, devaluations and revaluations are still arranged in terms of the ECU, but this has no economic significance. It is impossible *ex ante* to define all central rates in terms of the ECU.

To make possible the use of Community currencies for compulsory intervention at the margins, the central banks of the countries participating in the Exchange Rate Mechanism (ERM) of the EMS arrange mutual very-shortterm credit facilities (VSTF) in unlimited amounts. (These countries are Belgium, Denmark, France, Germany, Ireland, Italy, Luxembourg, and the Netherlands. The United Kingdom and Greece are part of the EMS but not of the ERM.) Spot sales and purchases of Community currencies are credited or debited against the central banks' ECU accounts with the EMCF as of the "value date," or day of intervention. In this way, "secondary ECUs" are created in the form of net creditor and net debtor positions.

A financing operation falls due for repayment forty-five days after the value date, with the possibility of an automatic three-month extension of the settlement date at the request of the debtor central bank. If settlement is not made by means of holdings in the creditor's currency, it is accomplished entirely or in part by transferring "primary ECUs," with the proviso that a creditor central bank is not generally obliged to accept ECUs in settlement of more than 50 percent of its claim.<sup>2</sup> If settlement is only partially effected through ECU

<sup>&</sup>lt;sup>2</sup> Note that the EC central banks can hold working balances in Community currencies within

transfers, the balance is settled by transferring other reserve assets in accordance with the composition of the reserves of the debtor central bank.

While the official ECU has had some importance in the actual operation of the EMS, it has not played the pivotal role some originally envisaged. It has been confined to serving as an official reserve asset and as a unit of account for the VSTF and settlement. Even in this respect, its use has been somewhat limited, largely because dollar interventions have predominated, accounting for two-thirds of total interventions, while mandatory marginal interventions have accounted for only about 10 percent (see Table 1 and Micossi, 1985, for a detailed analysis).

Since the general exchange-rate realignment of March 1983, recourse has not often been made to interventions at the margin and to the VSTF arrangement. To explain why, I must first draw a distinction between two types of intervention in Community currencies:

- Symmetric monetary-base interventions occur whenever the intervening central bank uses reserves held at the central bank of the country whose currency is being used, thereby causing simultaneous opposite movements in the monetary bases of the two countries. Interventions at the margin that resort to the VSTF mechanism or spot settlements in official ECUs have this property. Unless sterilized, they imply a contraction of the monetary base in the weak-currency country and an expansion in the strong one.
- Asymmetric monetary-base interventions occur when one central bank uses private-market reserves—Eurocurrency or domestic private banking assets—in the other currency. In this case, apart from absorbing or releasing commercial-bank reserves held against domestic deposits subject to reserve requirements, these interventions affect only the monetary base of the country that initiates them, not of the country whose currency is being used.

Since the 1983 realignment, all central banks have sought by preventive intramarginal interventions to avoid the tensions that necessarily arise when exchange rates reach their bilateral margins. This has led some central banks to suggest extending use of the VSTF to finance intramarginal interventions. Any such automatic extension has been firmly resisted by strong-currency central banks, however; presumably they want to maintain control over their domestic monetary policies, which would be impaired by symmetric monetary-base interventions.

These contrasting preferences led to a compromise solution. A "mobilization clause" was introduced as part of a larger package adopted in 1985 to im-

limits laid down by the Committee of Governors. These limits may be exceeded only with the consent of the central banks concerned.

#### TABLE 1

|   | March 1979-<br>March 1983 | April 1983-<br>Dec. 1985 |
|---|---------------------------|--------------------------|
| U.S. dollars:                               |                           |                          |
| Purchases                                   | 34.8                      | 18.8                     |
| Sales                                       | 107.0                     | 49.1                     |
| ERM currencies:                             |                           |                          |
| At the limits <sup>a</sup>                  | 28.0                      | 7.9                      |
| Intramarginal:                              |                           |                          |
| Purchases                                   | 11.8                      | 27.9                     |
| Sales                                       | 22.5                      | 15.7                     |
| Other currencies: <sup>b</sup>              |                           |                          |
| Purchases                                   | 0.1                       | 3.2                      |
| Sales                                       | 2.4                       | 0.7                      |
| Total:                                      |                           |                          |
| Gross                                       | 206.6                     | 123.3                    |
| Net <sup>c</sup>                            | -85.2                     | -15.6                    |
| Memorandum items:                           |                           |                          |
| Recourse to VSTF                            | 20.9                      | 3.4                      |
| Use of ECU for settlements of interventions | 6.9                       | 0.1                      |

# FOREIGN-EXCHANGE INTERVENTION BY COUNTRIES IN THE ERM (in billions of U.S. dollars)

<sup>a</sup> Interventions at the limits involve balanced purchases and sales; the two are therefore taken together.

<sup>b</sup> The figures for 1985 include some interventions in the private ECU market.

<sup>c</sup> A minus sign indicates net sales.

NOTE: According to BIS recording practices, intervention figures do not include operations such as customer transactions, swaps with commercial banks, and forward settlements or other transactions that constituted a significant source of positive changes in gross official reserves. This explains why the recorded total change is a high negative figure.

SOURCE: Updated figures from Micossi (1985).

prove the usability of the official ECU. In view of the potential importance of this package for expansion of the ECU in the private markets, I briefly describe its contents (for more detail, see Committee of Governors, 1985, Instrument of 10th June 1985):

a. The mobilization mechanism enables EC central banks in need of intervention currencies to mobilize their net creditor positions in the EMCF, together with some of the ECUs that the EMCF allocated to them against the deposit of gold and dollar holdings. The EC central banks have agreed to facilitate such mobilization operations by providing dollars in proportion to their outstanding ECU swaps with the EMCF (but not in excess of those swaps). The dollars provided in this way may be exchanged for EC currencies with the approval of the central banks issuing those currencies. Mobilization operations run for three months, are renewable for a further three-month period, and bear interest at market rates. In exceptional circumstances, a central bank may opt out partly or entirely from participation in such operations.<sup>3</sup>

b. The payments ratio limiting settlements in official ECUs of obligations arising from the use of the VSTF (the acceptance limit) remains at 50 percent as a general rule, but this limit can now be waived to the extent that the recipient central bank is itself a net debtor in ECUs.

c. The return on net positions in ECUs and on ECU-denominated claims under the VSTF has been improved. The interest rate was previously the weighted average of the official discount rates of the member countries; it is now a weighted average of representative money-market rates in those countries.

d. Central banks of nonmember countries and international monetary institutions such as the BIS, which are accorded the status of "other holder" by the EMCF Board, may obtain official ECUs from EC central banks by means of repurchase agreements or reversible swap transactions. "Other holders" are entitled to the same return on their ECU holdings that EC central banks receive on their net ECU positions.

### **3** The Rapid Development of the Private ECU

In March 1975, before the inception of the EMS, a new European Unit of Account (EUA) was introduced by the Community with a basket-type definition like that subsequently adopted for the ECU (see Allen, 1986, pp. 3-4). It was used in private financial markets as an indexation device for credit contracts (see Aschheim and Park, 1976, and Committee of Governors, 1985, pp. 73-75). But it was not until the inception of the EMS and the establishment of the ECU that its use grew substantially in these markets. The ECU market has shown remarkable growth in terms of both credit and exchange transactions. Because excellent reviews of recent developments are available (see, e.g., BIS, 1985, pp. 127-133, and Allen, 1986), I will deal briefly here only with what I regard to be the key features of this expansion.

## The Private ECU Banking Market

The private ECU banking market has grown very significantly compared with other currency sectors of the international banking market, especially from 1982 to 1985. By the end of 1985 it ranked fifth, with a share of more than 3 percent (Table 2). The expansion was mainly the result of borrowing by nonbank residents in Italy and France (see Table 3); expecting broad exchange-rate stability within the EMS, they were attracted by the lower inter-

 $^3$  The first activation of the mobilization scheme took place in December 1985 at the request of the Bank of Italy.

## TABLE 2

GROWTH OF THE ECU BANKING MARKET COMPARED WITH OTHER SECTORS OF THE INTERNATIONAL BANKING MARKET IN EUROPE: Amounts Outstanding, End 1982–End 1985 (in billions of U.S. dollars)

|                       | 1982               |                  | 19      | 1983                 |         | 184                  | 1985                 |                  |
|-----------------------|--------------------|------------------|---------|----------------------|---------|----------------------|----------------------|------------------|
| Currency <sup>a</sup> | Assets             | Liabil-<br>ities | Assets  | Liabil-<br>ities     | Assets  | Liabil-<br>ities     | Assets               | Liabil-<br>ities |
| U.S.\$                | 832.4              | 869.5            | 869.2   | 908.5                | 893.4   | 943.9                | 924.8                | 978.6            |
| DM                    | 155.9              | 141.6            | 150.0   | 135.6                | 142.9   | 132.6                | 204.1                | 191.2            |
| SwF                   | 79.0               | 71.5             | 77.0    | 70.7                 | 67.6    | 62.1                 | 103.1                | 92.8             |
| ¥                     | 30.6               | 31.8             | 28.9    | 33.4                 | 32.6    | 32.0                 | 71.6                 | 68.3             |
| ECU                   | $6.5^{\mathrm{b}}$ | $5.5^{b}$        | 11.9    | 10.0                 | 28.2    | 22.3                 | 54.2                 | 48.2             |
| £                     | 15.5               | 18.0             | 14.8    | 16.4                 | 16.6    | 17.6                 | 26.1                 | 29.1             |
| Other                 | 63.8               | 67.4             | 70.2    | 69.8                 | 63.9    | 64.7                 | 88.1                 | 94.9             |
| Total                 | 1,183.7            | 1,205.3          | 1,222.0 | $\overline{1,244.4}$ | 1,245.2 | $\overline{1,275.2}$ | $\overline{1,472.0}$ | 1,503.1          |

<sup>a</sup> See the Table of Currency Symbols on p. 34.

<sup>b</sup> Estimated.

NOTE: Domestic and foreign positions of banks in Europe only.

SOURCE: BIS quarterly statistics on international banking developments.

| TABLE 3  |
|--|
| STRUCTURE OF THE ECU BANKING MARKET: AMOUNTS OUTSTANDING AT END-1985 |
| (in billions of ECUs)  |

|                       |         | Luxem-     |        |       | Nether-    | United  | Other       |       |                   |
|-----------------------|---------|------------|--------|-------|------------|---------|-------------|-------|-------------------|
|                       | Belgium | bourg      | France | Italy | lands      | Kingdom | Countries * | Total | M.I. <sup>b</sup> |
| Assets of banks from: |         |            |        |       |            |         |             |       |                   |
| Nonresidents:         |         |            |        |       |            |         |             |       |                   |
| Banks                 | 6.9     | 3.2        | 8.5    | 4.7   | 3.1        | 7.5     | 1.1         | 35.0  | 31.1              |
| Nonbanks              | 1.4     | 1.8        | 1.5    | _     | 0.5        | 1.5     | 1.1         | 7.8   | 6.9               |
| Residents:            |         |            |        |       |            |         |             |       |                   |
| Banks                 | 2.0     | 1.8        | 4.3    | 0.8   | 0.3        | 2.7     | 0.2         | 12.1  | 10.8              |
| Nonbanks              | 0.1     | 0.2        | 1.7    | 2.7   | 0.2        | 0.5     | <u>0.7</u>  | 6.1   | 5.4               |
| Total                 | 10.4    | 7.0        | 16.0   | 8.2   | 4.1        | 12.2    | 3.1         | 61.0  | 54.2              |
| Liabilities of        |         |            |        |       |            |         |             |       |                   |
| banks to:             |         |            |        |       |            |         |             |       |                   |
| Nonresidents:         | *.      |            |        |       |            |         |             |       |                   |
| Banks                 | 5.6     | 2.6        | 7.4    | 7.6   | 1.4        | 8.2     | 2.3         | 35.1  | 31.2              |
| Nonbanks              | 0.8     | 1.3        | 0.5    | 0.2   | 0.6        | 0.2     | 0.1         | 3.7   | 3.2               |
| Residents:            |         |            |        |       |            |         |             |       |                   |
| Banks                 | 1.9     | 2.0        | 4.3    | 0.8   | 0.3        | 2.4     | 0.2         | 11.9  | 10.6              |
| Nonbanks              | 0.3     | <u>0.8</u> | 0.1    |       | <u>1.5</u> | 0.6     | <u>0.3</u>  | 3.6   | 3.2               |
| Total                 | 8.6     | 6.7        | 12.3   | 8.6   | 3.8        | 11.4    | 2.9         | 54.3  | 48.2              |

<sup>a</sup> Austria, Denmark, Germany, Ireland, Spain, and Sweden.

<sup>b</sup> Memorandum item in billions of U.S. dollars.

NOTE: "Banks" includes other entities not explicitly identified as nonbanks by the banks reporting to the BIS.

SOURCE: BIS quarterly statistics on international banking developments.

est cost of ECU borrowing. It can be estimated that at end-1985 Italian enterprises accounted for ECU 5.4 billion of the ECU 13.9 billion in total loans granted by banks to nonbanks, of which ECU 2.7 billion were from Italian banks and ECU 2.7 billion from banks located in other countries. The corresponding figures for French enterprises were ECU 2.4 billion, of which ECU 1.7 billion were from French banks and ECU 0.7 billion from other banks. Italian and French final borrowers thus accounted for 55 percent of total ECU bank loans to nonbanks.

On the deposit side, funds have traditionally come primarily from Belgium and Luxembourg. It is worth noting, however, that in 1985 Dutch residents were major suppliers of funds—an important development, since it refutes the argument that the ECU cannot be attractive to residents of strong-currency countries. Indeed, there is no reason why the expectation of relative intra-EMS exchange-rate stability that probably led Italian and French residents to borrow in ECUs should not induce Dutch and German residents to lend in ECUs.<sup>4</sup> At end-1985 banks in Belgium and Luxembourg accounted for ECU 3.2 billion (44 percent) of total nonbank deposits in ECUs, but nearly two-thirds of these deposits came from nonresidents, mainly of Dutch and German nationality. If an estimated ECU 0.5 billion of Dutch deposits with banks domiciled in Belgium-Luxembourg is added to the ECU 1.5 billion of Dutch deposits held in domestic banks, the total amounts to nearly 30 percent of the ECU bank-deposit base.

To put the expansion of the ECU market into perspective, a comparison should also be drawn between the ECU-denominated assets and liabilities of Community nonbank residents and their overall assets and liabilities vis-à-vis banks. As Table 4 shows, in spite of the recent very rapid growth of the private ECU, ECU loans and deposits still represent only a small proportion of total liabilities and assets—0.7 and 0.2 percent respectively.

In all Community countries, with the notable exception of Germany, the ECU is treated—*de jure* or *de facto*—as a foreign currency. ECU transactions are permitted, but they fall under foreign-exchange restrictions against capital outflows, which, although declining, still exist in Denmark, France, Greece, Ireland, and Italy. Thus the controls impede the ability of residents to establish ECU deposits with banks in other countries.

Capital movements are unrestricted in Germany, so its citizens can acquire ECU deposits with banks not domiciled in Germany. Nevertheless, Article 3, paragraph 2, of the Currency Act prohibits residents from entering into indexed debts unless explicitly authorized by the Bundesbank, and the ECU falls under this provision, because it is treated in Germany as a unit of

<sup>&</sup>lt;sup>4</sup> Market operators point out also that unless a realignment is anticipated in the short term, certain categories of savers in strong-currency countries show a preference for nominal yields higher than those available on their domestic currencies.

#### TABLE 4

|          | Outst          | anding Borroy    | wing                      | Outs             | tanding Depos               | sits                      |
|----------|----------------|------------------|---------------------------|------------------|-----------------------------|---------------------------|
|          | In ECUs<br>(1) | Overall ª<br>(2) | (1) as %<br>of (2)<br>(3) | In ECUs<br>(4)   | Overall <sup>b</sup><br>(5) | (4) as %<br>of (5)<br>(6) |
| Domestic | 77             | 1 488 2          | 0.52                      | 1.0              | 1 465 7                     | 0.12                      |
| Foreign  | 1.1            | 1,400.0          | 0.52                      | 1.9              | 1,403.7                     | 0.13                      |
| banks    | 3.0            | 155.3            | 1.93                      | 1.3              | 71.6                        | 1.82                      |
| Total    | 10.7           | 1,643.6          | 0.65                      | $\overline{3.2}$ | 1,537.3                     | 0.21                      |

Community Nonbank ECU-Denominated Assets and Liabilities Vis-a-Vis Banks and Their Overall Supply and Recourse to Bank Credit Markets: Amounts Outstanding at End-March 1985 (in billions of ECUs)

<sup>a</sup> Domestic bank lending to the private sector in EC countries and international bank lending to EC residents (excluding Greece).

<sup>b</sup> Broad money stock in EC countries and externally held bank deposits of EC residents.

SOURCE: BIS (1985) and quarterly statistics on international banking developments; Committee of Governors of the EEC Central Banks, monthly statistics.

account rather than a foreign currency. Accordingly, German banks cannot issue ECU deposits to residents and German residents cannot incur ECU liabilities.<sup>5</sup> The main objections commonly raised by the Bundesbank to declaring the ECU to be a currency or placing it on an equal footing with a currency are as follows: (a) the ECU is not backed by any independent monetary authority responsible for its internal and external value; (b) there is no guarantee of the ECU's continuity of value as long as changes in the weights in the basket can be decided upon by the competent authority; (c) there is no institution or arrangement ensuring ready convertibility of the ECU into reserve currencies; (d) there is no clear way to establish a role for the ECU as an intervention currency. A strict line on these points is reported by Whalig (1985), and a somewhat more open one by Poehl (1985). For an opposite view

<sup>5</sup> The Currency Act was passed in 1948 as part of the currency reform carried out by the occupying powers with a view to establishing and guaranteeing the DM's monopoly as legal tender. Under the Currency Act, residents must be authorized by the Bundesbank if they wish to use currencies other than the DM. With the introduction of the Foreign Trade and Payments Act in 1961, residents were allowed to enter into foreign-currency commitments with nonresidents. In that year, moreover, the Bundesbank issued a general authorization permitting foreign-currency commitments between residents. This general authorization does not apply to the ECU, which is governed by the regulations on indexation clauses. Over the years, in the interest of a stabilityoriented monetary policy, the Bundesbank Central Council has always avoided giving authorizations of this sort in connection with money or capital transactions. and the legal arguments supporting it, see Carbonetti (1987) and Harlandt (1986). See also the discussion of these points in section 7 below.

Largely in view of these institutional factors, the private ECU market continues to be dominated by nonbank *borrowers*. As can be seen from Table 3, only about half of total ECU loans granted by banks to nonbank final borrowers (ECU 13.9 billion) are accounted for by nonbank deposits (ECU 7.3 billion). The banking system covers the rest (a) by borrowing the corresponding basket currencies directly, (b) by borrowing a single currency and simultaneously covering in the forward market, or (c) by borrowing private ECUs from central banks, which hold ECU bank deposits as foreign-exchange reserves.<sup>6</sup>

These factors and the complex web of lending and borrowing that links final savers and users of funds help explain the importance of interbank transactions. Interbank claims and liabilities in other currency sectors of the international banking market correspond to some 70 percent of overall claims and liabilities. Interbank claims and liabilities in ECUs comprise about 75 percent on the assets side and 85 percent on the liabilities side.

## The Private ECU Bond Market

In 1985, for the first time, the ECU rose to fifth place among currencies in which foreign bond offerings were denominated (Table 5). Italy, France, and the EC institutions were the largest borrowers. In Italy, especially, ECU issues to residents have become significant. In 1984 and 1985, however, there was increasing participation by non-EC countries and institutions, and notably by operators based in the United States and Japan.<sup>7</sup> In 1982, over 90 percent of total ECU issues were accounted for by EC borrowers; by 1985, the share had declined to some 60 percent.

## The Private ECU Exchange Market

The private ECU exchange market has also expanded significantly. The aggregate daily turnover in Community countries can be roughly estimated at ECU 2.5 to 3 billion. Of this, Belgium and the United Kingdom account for ECU 0.8 to 1 billion and Italy for 0.5 billion. Turnover in Denmark, France, and the Netherlands is between ECU 0.2 and 0.4 billion. In every country except Italy, interbank transactions exceed trade-related transactions.

<sup>6</sup> The perception of many market operators is that the imbalance is progressively narrowing, not only because of central-bank deposits but also because of the issue of new instruments, such as certificates of deposit, to raise funds from nonbanks directly in ECUs.

<sup>7</sup> While Japanese borrowers are tapping the market for genuine diversification purposes, it appears that prime U.S. borrowers have been attracted to the ECU market mainly to benefit from the marginally lower cost they enjoy compared with European borrowers. The ECU funds raised are then often swapped against dollars with European borrowers.

#### TABLE 5

| Currency <sup>a</sup> | 1982  | 1983  | 1984  | 1985  |
|-----------------------|-------|-------|-------|-------|
| U.S.\$                | 63.9  | 57.0  | 64.2  | 61.0  |
| SwF                   | 15.0  | 17.5  | 11.8  | 8.9   |
| ¥                     | 5.0   | 5.3   | 5.4   | 7.7   |
| DM                    | 7.1   | 8.6   | 6.0   | 6.7   |
| ECU                   | 1.1   | 2.8   | 2.6   | 4.2   |
| £                     | 2.6   | 3.9   | 5.0   | 4.0   |
| A\$                   | •••   | 0.3   | 0.3   | 1.8   |
| Can\$                 | 1.6   | 1.4   | 2.0   | 1.7   |
| DGlr                  | 1.1   | 1.2   | 0.8   | 0.6   |
| Other                 | 2.6   | 2.0   | 1.9   | 3.4   |
| Total                 | 100.0 | 100.0 | 100.0 | 100.0 |
| Memorandum item:      |       |       |       |       |
| Total issues (in bil- |       |       |       |       |
| lions of U.S.\$)      | 75.5  | 77.2  | 111.5 | 167.8 |

# CURRENCY DISTRIBUTION OF EXTERNAL BOND OFFERINGS (in percent)

<sup>a</sup> See the Table of Currency Symbols on p. 34.

SOURCE: Organization for Economic Cooperation and Development.

As a result of the growth of the ECU exchange market, spreads between buying and selling rates of the Community currencies compared with the ECU are, on average, not too different from those recorded by the Community currencies compared with the dollar. The ECU is now quoted officially in Amsterdam, Athens, Brussels, Copenhagen, Lisbon, Milan, Oslo, Paris, and Rome. Forward markets are still limited, however, although future contracts in ECUs have been launched in Chicago and New York, and options have recently been introduced.

A potentially important development for the private ECU exchange market is the previously mentioned involvement of some EMS central banks (see Table 6). Before 1985, interventions in private ECUs (i.e., market purchases or sales of ECUs against domestic currencies) were prompted by the desire to maintain orderly conditions in the ECU market. In 1985, however, interventions also aimed in some instances at stabilizing the domestic currency. Total interventions amounted to some ECU 2.4 billion on a gross basis, the bulk of which were conducted by the Bank of Italy.

In December 1985, net holdings of private ECUs by Community central banks amounted to 1.7 billion, or 2 percent of their foreign-exchange holdings. One-third of these reserves were held by the Bank of Italy, accounting for 5 percent of Italy's total foreign-exchange reserves. The structural imbalance between final borrowers and final lenders in the ECU market is thus partly offset by central-bank deposits.

#### TABLE 6

|                                       | Total Europea | in Community | Ita   | y     |
|---------------------------------------|---------------|--------------|-------|-------|
|                                       | 1984          | 1985         | 1984  | 1985  |
| Purchases against                     |               |              |       |       |
| domestic currency a                   |               |              |       |       |
| On the Market                         | 61            | 1,530        | 57    | 275   |
| Off the Market                        | 1,078         | 1,589        | 898   | 1,071 |
| Sales against                         |               |              |       |       |
| domestic currency <sup>a</sup>        |               |              |       |       |
| On the market                         | •••           | 870          | •••   | 759   |
| Off the market                        | 43            | 165          | 25    | 103   |
| Purchases against                     |               |              |       |       |
| foreign currency                      | 33            | 151          | •••   | 50    |
| Sales against                         |               |              |       |       |
| foreign currency                      | 1,026         | 1,358        | 819   | 760   |
| Other transactions (net) <sup>b</sup> | 36            | -177         | 25    | •••   |
| Increase-decrease                     |               |              |       |       |
| in ECU net holdings                   | 130           | 708          | 136   | -226  |
| Net holdings at                       |               |              |       |       |
| end of period                         |               |              |       |       |
| In absolute terms                     | 953           | 1,662        | 776   | 550   |
| % of foreign-                         |               |              |       |       |
| exchange holdings                     | 1.12%         | 2.22%        | 4.49% | 5.10% |

#### Central-Bank Transactions and Holdings in Private ECUs, January 1984–December 1985 (in millions of ECUs)

<sup>a</sup> Purchases and sales by each central bank of private ECUs against domestic currency on the market (i.e., interventions) and off the market (e.g., transactions with customers or uptake of the proceeds of government borrowing abroad).

<sup>b</sup> Other receipts and payments in private ECUs (e.g., interest received and paid on outstanding assets and liabilities in private ECUs).

SOURCE: Committee of Governors of the EEC Central Banks.

Recall that private-ECU interventions have "asymmetric" monetary-base properties, in contrast to official-ECU interventions. Furthermore, since a purchase or sale of ECUs is also a purchase or sale of the domestic currency, to the extent of its weight in the basket, ECU interventions must be larger than other interventions to achieve the same exchange-rate effect.

## 4 Reasons for the Growth of the Private-ECU Market

A number of explanations have been advanced for the growing use of the ECU by private operators. In my view, the explanation must be sought in the synergistic interaction of these factors: (a) the basket properties of the ECU; (b) the effects of the exchange-rate and economic-policy commitments made

by the countries participating in the ERM; and (c) the general process of financial innovation (the latter point was made by Poehl, 1985, and convincingly developed by Levich, 1986).

For many European operators, the ECU's total yield—exchange rate plus own interest—reduces exposure in international asset and liability management. At the same time, the ready-made basket reduces transactions costs compared with the costs of transactions in individually mixed currency cocktails.

Since the inception of the system, participants in the ERM with traditionally weak currencies have generally more than offset the expected depreciation of their currencies by offering higher nominal interest rates. Thus, particularly from March 1983 to July 1985, importers and exporters in Italy and France were attracted to the ECU as an instrument for international borrowing by its relatively low interest rate and its limited anticipated appreciation against the domestic currency. By contrast, savers in the Benelux countries were moved to buy ECU bonds by yields higher than those available on domestic bonds and by the lower exchange-rate risk compared with the acquisition of assets in high-interest currencies. Moreover, residents of France and Italy seeking portfolio diversification found that purchases of ECU bonds were a partial substitute for purchases of assets in foreign currencies, which were discouraged by foreign-exchange restrictions.

There is some truth to the argument that the growth of the private ECU owes much to the uncertainty about Community exchange and interest rates arising from its unsatisfactory progress toward monetary integration, but this should not be overrated. It is no coincidence that the years 1982-85—a period of relative stability in intra-EMS exchange rates and sharp fluctuations in quotations against the dollar—saw the most significant expansion of the private-ECU market.

The explanation of the growth of the private-ECU market can be subjected to empirical analysis by examining the role of the ECU in a rational portfolio strategy for the European resident engaged in international financial transactions.<sup>8</sup>

First, the usual mean-variance model can be used to assess the relative attractiveness of the various currencies. In this approach, a meaningful comparison must be made sequentially, by taking each currency in turn as the "domestic" currency and comparing each foreign currency as a mutually exclusive option. (The variability of the domestic currency against itself reflects the variability of the domestic interest rate and is typically much lower than

<sup>&</sup>lt;sup>8</sup> Important technical contributions along these lines have been made by Hamaui (1985) and Jorion (1986). As Ciampi (1981) explains, the fixed-quantities definition of the official ECU was chosen to allow private operators to replicate the official basket, thereby fostering private use of the ECU and exploiting its portfolio properties.

the variability of a foreign currency, which includes the variation in the relevant exchange rate.)

The analysis covers the entire period of floating, March 1973 to December 1985. Two subperiods are also considered, March 1973 to March 1979 and the period since the inception of the EMS. In order to extend the analysis to the entire period, the present currency definition of the ECU is carried back to 1973. In addition to the ECU and all its component currencies, the U.S. dollar, the Japanese yen, and the SDR basket on its 1985 definition are also considered. Average data for interest rates, exchange-rate variations, and total returns are computed on a quarterly basis; interest rates refer to domestic money-market instruments.

Over the entire period of floating, the DM, the dollar, and the yen showed approximately the same return (see Figure 1). As to currency baskets, the SDR performed better than the ECU, mainly as the result of two long waves in the dollar exchange rate, with a low tide in the first half of the period and a high tide in the second half, which was broadly contemporaneous with the EMS experience (Figure 2). The two long waves are broken by strong crosscurrents: the dollar-appreciation phase came to an end in February 1985, and a significant decline took place thereafter. These developments are summarized in quantitative terms in the Appendix. Tables A.1a and A.1b report both the interest-rate and the exchange-rate components. They also show the standard deviations of recorded returns.







FIGURE 2 Compounded Return in Dollar Terms of Money-Market Investments in Five Currencies, 1979I-1985IV

If we concentrate our attention on reward-to-variability ratios of foreign investments during the period of operation of the EMS, some interesting features emerge. First, taking in turn as the domestic currency the DM, the French franc, and the Italian lira, the ECU represents the "best" investment opportunity in each case. Taking the Dutch guilder as the domestic currency, the ECU is "second best," closely following the DM (Table A.1b). For the U.K. or U.S. investor, on the other hand, the most satisfactory choice—still on the basis of the reward-to-variability ratio—is the SDR. But the SDR shows up as a very poor choice for all investors whose currencies participate in the ERM.

A more satisfactory analysis of international financial portfolios can be made by allowing for the covariance among alternative returns (Solnik, 1973, 1974; Adler and Dumas, 1983). The standard Markowitz (1952, 1959) model can be used to determine the weights of currencies in "efficient" portfolios, with a view to minimizing the overall variance for a given expected return. In this case, too, the exercise is carried out sequentially, taking each currency in turn as the domestic currency and considering all other currencies as elements of the "foreign" portfolio. The parts of the portfolio invested in foreign short-term assets are constrained to be positive (an unconstrained approach is followed by Hamaui, 1985, and Jorion, 1986), ruling out short sales. The viewpoint is that of an investor rather than a borrower (negative weights) or an arbitrageur (negative or positive weights). Consequently, lending rather than borrowing rates are considered. Eliminating constraints on the signs of the weights would have required assuming equal lending and borrowing rates and an unbounded objective function (i.e., unbounded theoretical profits) with very large offsetting positions in highly correlated currencies. Such assumptions would have led often to empirical results of little economic importance.

For the period of operation of the EMS—March 1979 to December 1985 the simulations of efficient portfolios for investors in different countries again show that the ECU represents a good investment for European residents. It appears, in combination with other currencies, in the efficient-portfolio frontier of all EMS residents (Table A.2b). Moreover, it also lies close to the frontier as a low-risk asset. This is true for residents of both weak- and strongcurrency countries (Figures A.1 and A.2). The fact that the ECU is always present with significant weights in minimum-variance portfolios demonstrates its special attractiveness for risk-averse operators. The weights vary from 0.98 for the French investor to 0.24 for the German one. A combination of ECUs (0.77) and SDRs (0.23) is the efficient least-variance portfolio for U.K. residents.

When the dollar and the yen are taken as base currencies, the portfolio analysis points to the desirable properties of the SDR, which for the non-European investor performs a role similar to that of the ECU for the European investor. These results suggest that there is something to the "infant currency" argument. Without (a) the active initial support of the EC Commission and of some European countries and (b) the use of the official ECU, it might have been difficult to reach the threshold at which the private ECU could assert itself in the markets. Now that the private-ECU market exists, substantial benefits can be reaped from the significant reduction in transactions costs compared with those of individually tailored baskets and the SDR itself.

## 5 An Assessment of the Present and Future Status of the ECU

We have seen that the ECU has perhaps fallen short of expectations in the official field but has run ahead of them in the private markets. Although technically the EMS could function even without the ECU, emphasis continues to be placed on assigning the ECU a central role in the system, especially by European political forces.

Advocates of this emphasis insist on two points: (a) the ECU should be developed into a full-fledged international reserve asset, eventually competing with the dollar; (b) the ECU should gradually acquire the status of a true European currency, functioning parallel to national currencies in the initial phase and eventually becoming the common currency. Objective (a) is usually related to developments in the official field, objective (b) to advances in private markets.<sup>9</sup>

This continuing interest is indicated by the recent reference to the ECU in the amended Treaty of Rome—the Constitutional Charter of the EC. The objective of monetary cooperation was explicitly recognized in the revision of the texts that was approved in January 1986. The Preamble of the Single Act referred to the fact that in 1972 the heads of state had approved the goal of progressive realization of economic and monetary union, and a new Article (102 A) was introduced in the Treaty: "In order to ensure the convergence of economic and monetary policies which is necessary for the further development of the Community, Member states shall . . . take account of the experience acquired in cooperation within the framework of the EMS and in developing the ECU . . ." (European Single Act, Brussels, 1986).<sup>10</sup>

Among the many reasons for promoting the ECU's role as an international reserve asset is that it would reduce European dependence on the dollar, allowing easier decoupling of policies if necessary. In a world characterized by floating exchange rates between major currencies and by EMS exchange-rate agreements, cohesion among the EC currencies would reduce the impact of dollar swings.

To achieve cohesion, intra-EC international reserves are needed. The demand might be satisfied by national currencies, but problems would be likely to arise for one country, such as Germany, if its currency began to serve as the main reserve currency and its liquid liabilities to foreigners grew disproportionately. Swings of confidence could put a severe burden on domestic monetary policy. The availability of the ECU as an alternative to the DM might lessen these difficulties and prevent the emergence of EMS tensions as a consequence of switches out of the dollar. These positive effects would be enhanced if the ECU were also used as an instrument for invoicing international transactions and pricing international commodities.

If the ECU is to play an important role as an international reserve asset, it must be made more fully usable as a means of international payments. While this statement may appear trivial, the point is that there is no inconsistency between the development of the ECU in the official and private spheres. On the contrary, it is impossible to advance one use of the ECU beyond a critical point unless both aspects are brought together in a complementary way. The ECU cannot become an effective international reserve asset unless it is used

<sup>&</sup>lt;sup>9</sup> The latter distinction could be reversed, however, under present arrangements: the private ECU arguably has more of the attributes of a reserve currency than the official ECU.

<sup>&</sup>lt;sup>10</sup> To underline the relevance of the ECU, Sarcinelli (1985) argues that the difference between the snake and the EMS is the presence of the ECU. The same argument was made by the former Governor of the Bank of France, Michel Camdessus.

both by central banks and by private holders, particularly commercial banks.

Market intervention in ECUs, invoicing and pricing in ECUs,<sup>11</sup> and holding reserves in ECUs would represent the three activities necessary to establish the ECU as a full currency in both the official and private sectors: as a unit of account, medium of exchange, and store of value.<sup>12</sup> Uncertainty in the system would be reduced, in part as an automatic result of the fact that the vehicle currency is an average of participating currencies.<sup>13</sup>

It follows from the arguments developed so far that the most appropriate way to increase use of the official ECU is not to make its use mandatory in intra-EMS transactions, as by requiring the use of ECUs in settlements of intramarginal interventions or eliminating the present 50 percent payments ratio. Both would have "symmetric" monetary-base effects on surplus and deficit countries alike. This may be desirable once conditions of monetary stability have been achieved in all EMS countries, but not necessarily in the present transitional phase, when success is still uneven. A more fruitful approach would be to ensure transferability and convertibility of the official ECU by facilitating interchangeability between the official and the private sectors.

# 6 The ECU as Means of International Payments: A Suggested Scheme

Reserve currencies must be liquid and fully convertible into other currencies, represent a stable store of value, and provide a competitive return. The "portfolio" results reported in section 4 suggest that the ECU basket has some of these properties. In view of the enlargement of the EC, it would be better if the basket consisted only of currencies participating in the ERM; otherwise, the ECU would be weakened by the lack of policy commitment to exchange-rate stability.

However, the institutional aspects of the ECU discussed in sections 1 and 2 indicate that the official ECU also has important defects in terms of transferability and convertibility. Admittedly, acceptance limits were introduced

<sup>11</sup> On the general question of the choice of an invoice currency in international transactions, see Bilson (1983). As to the specific role of the ECU, see *ECU Newsletter* (1986). In the case of the ECU, the development of deep and resilient forward markets is important. Richard Mikkelsen and Herring Dalgaard have pointed out to me a key reason why the ECU has been slow to replace other currencies for invoicing: a company wanting to minimize its foreign-exchange risk in international trade can easily do so by using one of the major currencies for invoicing and then covering the entire risk in the forward market; using the ECU for this purpose only partially reduces exchange risk.

<sup>12</sup> For early proposals along these lines, see Ossola (1971) and Magnifico (1973).

<sup>13</sup> See Williamson (1983) for a forceful elaboration of these points with reference to the SDR, which *mutatis mutandis* can also be applied to the ECU and to European trade, trade financing, and reserve holding.

as protection against the risk of excessive ECU creation resulting from the automatic link to the price of gold. Today, however, those limits restrict the usefulness of the ECU as a reserve asset: central banks must be free to use the official ECU to intervene in the market. The main problem, however, is to establish the appropriate links between official ECUs and those traded in the market.

The problem and its possible solutions are not new to the theory of international finance. They have already been addressed in discussions of the role and expansion of the SDR, notably in contributions by the Bank of England in the Committee of 20 (IMF, 1974), by Coats (1982), and by Kenen (1983). These discussions provide a framework that can be adapted to the current problem of the ECU. The package of improvements to the use of the official ECU, described in section 2, will make matters easier, because the BIS has been designated an "other holder" of official ECUs. Three approaches are suggested to achieving transferability of the ECU between the official and private sectors:

a. The most direct way to achieve transferability would be to empower the EMCF to open accounts in *official* ECUs for private institutions, particularly commercial banks. This approach has obvious drawbacks, legal and institutional as well as economic. According to present legal provisions, official ECUs are created through three-month revolving swaps with the EMCF against the deposit of 20 percent of the gold and dollar reserves held by EMS central banks. The total amount of primary official ECUs is thus determined on an *ex ante* basis. Furthermore, only Community central banks and other designated official monetary agencies can be holders of official ECUs. To change these legal provisions would prove extremely difficult. A Council Regulation and consultation with the European Parliament would be unavoidable.

These difficulties apart, serious objections can be raised on strictly economic grounds. To start with, the total amount of high-powered ECUs would be affected by operations undertaken on the initiative of the private sector, and this would be unacceptable to EC central banks. EMCF operations would become extremely difficult. And serious problems would be encountered in deciding which private agents would be empowered to hold official ECUs.

A variant of this approach would allow commercial banks to hold official ECUs on the books of central banks. The drawback here is that private international interbank transactions could be cleared only through transfers of official ECUs by central banks. This would have the symmetric monetary-base consequences that are regarded as undesirable by certain EC monetary authorities under present conditions. b. A second line of attack would be to allow central banks to deposit *private* ECU balances with the EMCF. This scheme appears to be most directly aimed at widening the scope for use of official ECUs: through the intermediary of the EMCF, EC central banks and other official monetary agencies could more easily engage in swaps between private and official ECUs. It would not solve the problem of ECU convertibility, however, since any such swap operation would always be subject to mutual consent by two monetary authorities.

c. The third model—which is by no means inconsistent with the second is based on linking the private and the official markets through the intermediary of a clearinghouse, recognized as an "other holder" of official ECUs. This scheme, which was originally suggested by Kenen (1983) for the SDR, would work along the following lines. Central bank A wishes to acquire foreign exchange with a view to intervening to support its currency before it reaches the compulsory margin. A triangular operation is activated with a commercial bank in country B and the clearinghouse. Commercial bank B receives official ECUs sold by central bank A; since it cannot hold them, it must deposit them simultaneously with the clearinghouse in exchange for an instrument of deposit. Therefore, at the end of the operation, central bank A has acquired a private-ECU bank deposit with commercial bank B, and it is readily usable for intervention purposes. The central bank has lost official ECUs, which are on the books of the clearinghouse as an asset with the EMCF. Commercial bank B has recorded an increase in assets (ECU-denominated deposits with the clearinghouse) and an increase in liabilities (ECU-denominated deposits owed to central bank A). The clearinghouse holds an ECU asset with the EMCF and has an ECU liability to commercial bank B.

Further transactions between commercial bank B and other commercial banks can of course affect the ownership of deposits with the clearinghouse, but not the ownership of official ECUs. Note also that intervention sales by central bank A of the private ECUs thus acquired would imply the usual asymmetric monetary-base consequences: they would be concentrated on country A itself.

An advantage of this scheme is that it would not require any legal change. A joint decision by the Board of EC Governors, the Board of the EMCF, and the Board of the BIS would, in my opinion, be quite sufficient to make it operational. As mentioned earlier, the BIS has now been officially designated as an "other holder" of ECUs. In addition, the previous ECU clearinghouse system MESA (Mutual ECU Settlement Account), comprising seven commercial banks (Crédit Lyonnais, Lloyds, Banque Bruxelles-Lambert, Kredietbank-Brussels, Kredietbank-Luxembourg, Société Générale-Bruxelles, Istituto Bancario San Paolo), has been strengthened under the auspices of the BIS, which now acts as the final clearing institution.<sup>14</sup> Finally, the EMCF and the BIS operate under the same roof, which would facilitate the implementation of the scheme suggested here.<sup>15</sup>

Some key features of the new ECU clearing system have a bearing on the workings of the scheme proposed here. The projected system is characterized by two functions: (a) a netting function, which would record payment orders and the statement of net balances for each clearing bank vis-à-vis all other banks and would be entrusted to SWIFT (Society for Worldwide Interbank Financial Telecommunications), and (b) a banking function, which would work out the net balance for each clearing bank and its settlement, with the BIS playing the role of banking agent.

This clearing system is structured so as to avoid money creation: any clearing bank's negative clearance balance must be supported by liquidity. Since the ECU bank account opened by each clearing bank at the BIS can never be permitted to register a debit balance, each such bank must have sufficient liguidity in its BIS account to cover its net balance at the end of the day. The agreements with the BIS provide that the clearing banks' ECU accounts at the BIS be directly increased or decreased by channeling all of the ECU component currencies through the respective central banks. Liquidity thus consists, ultimately, of deposits of ECU component currencies in BIS bank accounts at the various central banks. All these currencies can thus be transformed through the system into ECU clearance balances. Should a clearing bank not be able to meet its debit balance when the BIS begins the clearing process at the end of a working day, the clearing for that day must be canceled. The clearing transactions undertaken by each participating bank are carried out on the following working day, but without any payment orders to and by the clearing bank in default.

The fact that private and official ECUs are already standardized makes it easier to implement the clearinghouse proposal. A potentially important problem is posed, however, by a difference in yield, since private ECUs earn Eurocurrency rates and official ECUs earn domestic money-market rates. But, given the present degree of financial deregulation, these differences are becoming smaller and should decline further; moreover, deposits with the BIS are likely to carry a security yield, because they have the security of BIS

<sup>14</sup> On these points, see BIS (1986). In March 1986, the ECU Banking Association and the BIS signed the convention for setting up the ECU clearing system. After a trial period, the system came into force at the end of 1986.

<sup>15</sup> An intermediate step in this direction would be to foster swaps of private and official ECUs between a central bank in need of private ECUs and the BIS, which performs the dual function of commercial bank and "recognized other holder." This exchange is already possible under existing rules (see Padoa-Schioppa, 1986).

backing, which may help offset the rate spread. The remaining gap would presumably have to be paid by the central bank initiating the operation (central bank A).

In the initial phase, various quantitative and qualitative limitations could be introduced to avoid excessive use of the proposed facility, the yield gap just referred to being in itself a limiting factor. These limitations could be lifted progressively as the necessary operating experience was gained. Some difficulties could also arise because of the precarious nature of the official ECU. This should not pose insurmountable problems, although the scheme is likely to result in pressure to consolidate the present three-month revolving swap system.<sup>16</sup>

A final possible problem with this scheme for linking official and private ECUs is analogous to the criticism made by Williamson (1983) concerning Kenen's (1983) original proposal for an SDR clearing union. Williamson argued that changes in the stock of SDR deposits at the clearinghouse should not be limited to the case of a central bank switching from official to private SDRs. He pointed out, however, that there would be a problem in maintaining the value of the SDR if commercial banks wished to increase their holdings of SDR deposits when no central bank was willing to reduce its holdings: the SDR's value would be bid up beyond that of the component currencies. Williamson therefore suggested that Kenen's scheme be extended to include an arrangement whereby the clearinghouse would also be empowered to issue SDR deposits in exchange for the basket of currencies, plus a small premium to cover the costs of the operation.

With respect to the ECU, the problem raised by Williamson is partly overcome through the workings of the private ECU market and of the new ECU clearing system. As we have seen, commercial banks are able to create the ECU deposits necessary to ensure balance in a basically lopsided market largely by "bundling" the component currencies into ECUs: this could be done through the intermediary of the BIS even now. In other words, it would be possible to rely on a market mechanism that would ensure that the value of the ECU deposits with the BIS remained in line with the value of the private ECU and the calculated value of the official ECU.

More generally, the question has to be addressed by taking proper account of the swap mechanism on which the creation of "primary" ECUs is based.

<sup>16</sup> It has been argued that the existing official ECU-creating mechanism is a major impediment to the full use of the ECU, not only because of the temporary nature of the swap operation but also because the amounts of ECUs vary in line with factors outside the Community's control (e.g., gold and dollar prices). While this is not the place to examine how arrangements could develop beyond the existing swap system, the problem should receive early attention. On this issue, see EC Monetary Committee (1978), Padoa-Schioppa (1980), Masera (1980), and Triffin (1984). The proposed mobilization mechanism, using a new BIS clearinghouse, would involve the creation by commercial banks of ECU term deposits. The maturity of these deposits would be less than (at the limit, equal to) the maturity of the official swaps between the central banks and the EMCF, which create primary ECUs.

Suppose the commercial banks that have acquired the official ECU deposits and transferred them to the BIS want to transfer them back to the central banks. Of course, when the original term contract matures, they can do this simply by reversing the original steps. Should the contracting parties agree to provide for operations to be canceled before maturity, the terms and the cost of such unwinding operations would have to be specified in the original contract itself.

Various options can be envisaged. Either the BIS would be able to execute the operation on its own account by borrowing alternative funds, or the BIS would turn to the central bank that originated the operation and ask it to convert the assets into private ECUs or any other currency. The central bank in turn could repay the BIS and ultimately the commercial bank in several ways: it could use its private ECU deposits, it could turn to the market and borrow private ECU or component currencies, or it could acquire foreign exchange from other central banks by activating the existing mobilization scheme for official ECUs.

# 7 Balance-of-Payments, Monetary, and Credit Implications of Increased Use of the Private ECU

I have contended that the ECU's development as an international reserve asset and, eventually, as a parallel common currency are complementary processes, contrary to what many believe. The emphasis on market convertibility of the official ECU must have as its counterpart increased use of the ECU as a monetary and credit instrument by private economic agents, in international as well as domestic transactions.<sup>17</sup> Therefore, it is necessary at this point to address these aspects of the process of ECU expansion in conjunction with preliminary comments on its balance-of-payments (exchange-rate) implications.

# **Balance-of-Payments Implications**

To what extent does the existence of the ECU market affect the volume and geographical pattern of international capital flows?<sup>18</sup> Clearly, if the existence

<sup>17</sup> Beyond private agents, official agencies—especially Treasuries—could also make more extensive use of the private ECU. The availability of short-term paper issued by prime borrowers such as Treasuries will be particularly important for giving the ECU the full character of an international currency.

<sup>18</sup> For a general analysis of this issue, see Mayer (1985).

and expansion of the ECU market have different credit and monetary implications for the various EC countries, there will be spillover effects on the respective balances of payments. Let me make three points.

First, note that in some instances the ECU serves *de facto* as an indexation device for domestic credit contracts. If the Italian government provides domestic residents with debt instruments indexed to the ECU, there are balance-of-payments consequences only to the extent that the newly supplied ECU-denominated assets satisfy residents' portfolio-diversification demands, relieving them of the need to acquire foreign assets through capital outflows.<sup>19</sup>

Second, because the ECU is a basket of currencies, its interest rate cannot diverge significantly from the weighted average of interest rates in the component currencies.<sup>20</sup> Accordingly, the ECU interest rate cannot play an important role in clearing the ECU deposit and credit markets, especially at the short-term end. As a spread between actual and theoretical rates arises, arbitrage forces will be set in motion that will lead to bundling or unbundling of the ECU into its component currencies and may thus lead to capital flows.

Third, and more generally, the foreign-currency denomination of credit contracts has no direct impact on the balance of payments of the country issuing the currency concerned, but it may have indirect effects. It is in fact likely that, as a consequence of the close links between the domestic market and the Euromarket for an international currency, residents of the vehiclecurrency country will be stimulated to enter into international transactions (to activate capital flows) that would otherwise not have taken place. With specific reference to the ECU, one implication is that a switch from the dollar to the ECU as a currency of denomination could reduce the elasticity of capital flows to and from the United States and foster greater intra-EC capitalmarket integration.

These considerations generally reinforce the argument that separate analysis of the effects of an increasing ECU role as a medium for international transactions and as a substitute for domestic currencies is bound to be futile, given the intimate relationship between the two processes.

In order to reach more specific conclusions, suppose now that an importer in country A decides to finance his trade by borrowing ECUs from the banking system, which we assume to be in balance. The excess demand for credits in ECUs does not bid up the ECU interest rate; instead, the banking system meets the demand by borrowing the component currencies to "bundle" them

<sup>&</sup>lt;sup>19</sup> If this is so, it enhances the argument for an expansion of ECU borrowing and lending between residents in countries that continue to restrict capital movements. On these points, see Modigliani and Capponi (1985).

<sup>&</sup>lt;sup>20</sup> I shall not take up the redefinition of the ECU here, except to express agreement with Triffin (1984, p. 56), who regards the issue as "revolutionary and premature at this stage, but imperative in the long, or even medium term."

into ECUs. The "autonomous," or voluntary, capital inflow into country A sets in motion compensatory, or involuntary, flows denominated in the currencies of the other EC countries and linked to their respective weights in the basket. These funds will in general be borrowed in the Euromarkets. This will trigger capital flows from the various countries involved to the extent that the pressures on the Eurocurrency markets spill over to the respective domestic money markets (i.e., according to the yield-elasticity conditions prevailing in each single Eurocurrency market).

In this way, the ECU banking market tends to influence the balance of payments of the countries issuing the component currencies of the basket. As we have seen, there is an element of automaticity in the process, mechanically linked as it is to currency weights in the ECU.

# Monetary and Credit Implications

On the basis of past experience, the overall yield of the ECU (interest adjusted for exchange-rate changes), combined with its variance and covariance, gives the ECU a place in the efficient portfolios of European residents. With the prospect of increasing monetary cohesion in the EMS, this should be even more the case in the future. As the risk of significant movements in intra-EC exchange rates declines, EC residents are increasingly likely to perceive the ECU as a closer substitute for the domestic currency than other international currencies, especially the dollar.

Shifts toward the ECU are particularly likely to occur when investors are confident that the ECU will provide a higher overall yield than the domestic currency. The converse would be true for borrowers. It might then appear that the market impetus will sometimes come from the lending side and other times from the borrowing side, with the ECU banking system covering through bundling operations. This is not, however, a foregone conclusion.

Take the following situation in countries with higher inflation and traditionally weak currencies: (a) these countries try to keep their exchange rates in line by offering real yields higher than those prevailing in lower-inflation countries; (b) exchange rates do not conform to purchasing-power parity but instead trend toward appreciation in terms of consumer-price inflation. Under these conditions, it turns out to be advantageous for the residents of these weak-currency countries to borrow in ECUs and for residents of strong-currency countries to lend in ECUs, rather than in domestic currency. Although it would prove even more advantageous under these circumstances to activate direct capital flows between the countries in question, the ECU riskcover properties explain why part of the flows may be intermediated by the ECU banking system: the ECU vehicle can provide a partial alternative to straight forward-cover operations.

Because the ECU can thus facilitate intra-EC capital movements, it increases the efficiency and integration of European credit markets while reducing the uncertainty costs for private operators. Inevitably, these welfare gains are achieved at the cost of some independence of domestic monetary policies. The shifts in intermediation from the domestic markets to the external ECU market also raise well-known questions about the possible repercussions on overall credit and money-creation processes.

The accepted view is that the questions must be addressed by analyzing the money- and credit-creating potential of Eurocurrency markets. The common wisdom is that these markets largely represent a substitute for, and not an addition to, domestic money and credit. Net additions are relatively low and result directly from the greater efficiency and monetary-base saving brought about by the growth of the Euromarkets. The mechanistic approach, which stressed the analogy with a domestic banking system, argued that the actual and potential multipliers were very high because of the low leakages into currencies and reserves. This approach has been discarded (see Swoboda, 1980, for a definitive paper on the argument).

As one of the first to develop a portfolio—as against a multiplier—approach to the workings of Euromarkets (Masera, 1972), I continue to stand by that line of thinking. I also agree with Allen (1986) that the portfolio approach provides the best explanation for the growth of the ECU market to date. But this is not the end of the story. With the prospect of growing Community use of the ECU, both domestically and internationally, there are some potentially important differences between the analysis of the ECU market and, say, the Eurodollar market.

Eurodollar balances have never had the potential to become an internal means of payment for residents of EC countries. Checking facilities have not been a significant feature of the Eurodollar markets; deposits, generally for very large amounts, have sometimes represented a temporary outlet for liquid funds, but they have always been closely linked to the financial transactions of international investors and traders. The absence of reserve requirements and the differential tax and supervisory treatment for Eurocurrency deposits, however advantageous, are not sufficient in themselves to activate a closed circuit of deposit formation and credit creation capable of becoming an alternative to domestic-currency banking.

The situation is different for the ECU. If use of the ECU becomes significant in domestic as well as foreign banking transactions, it could tend to supplant "controlled" bank operations. The ECU is in principle a close substitute for the domestic currency; if the advantages of Eurotransactions in ECU were open to ordinary banking transactions, problems of monetary control would emerge. This is one aspect of the general problem posed by the growing internationalization and deregulation of financial markets. The specific issue raised by the ECU is that it could—indeed should—become a better substitute for domestic currencies than any other "foreign" currency.

This argument is reinforced when central-bank redepositing in ECUs is

taken into account. It is commonly agreed, on the basis of analysis of the Eurodollar market, that central-bank redepositing of dollars in the Euromarket could trigger a multiple expansion of deposits, credits, and reserves. Take, for example, the case of a payments outflow from the United States and downward pressure on the dollar exchange rate. If central banks intervene to support the dollar and deposit the proceeds with the Eurobanking system, which in turn re-lends the funds outside the United States, an additional round of dollar interventions will ensue, and so the dance goes on.

A similar situation can arise in the case of the ECU under the following circumstances. Suppose that residents of country A find it cheaper to borrow in ECUs than in domestic currency. Starting from what is viewed as an equilibrium position by the monetary authorities of country A, the incipient capital inflow in ECUs would tend to put upward pressure on the domestic exchange rate (and downward pressure on domestic interest rates). The central bank in country A will intervene in the exchange market to offset the exchange-rate impulse and will sterilize the monetary-base impact of the intervention. If the proceeds of the foreign-currency purchase are re-lent to the ECU banking system, a chain can be activated whereby domestic borrowers build up external ECU liabilities that have as their counterpart "private" ECU assets held by the domestic central bank.

If we now permit residents to accumulate ECU deposits while avoiding reserve requirements and in addition make allowance for differences in risk behavior and in expectations about exchange rates, we can see how a process of credit and monetary creation could be set in motion that would add to, rather than substitute for, the creation of domestic-currency bank deposits and credit. If the ECU were to develop as a parallel domestic currency, it is not difficult to imagine how the competitive edge stemming from the lack of reserve requirements or other control devices on ECU banking in any single country would activate a monetary-credit circuit. In principle, this would have no balance-of-payments repercussions if the process took place in all EC countries, although a problem of credit and monetary control would arise. It is as if a system of nonbank intermediaries were created that offered deposits and credits fully competitive with those provided by the banks but imposed no reserve cost.

## 8 Conclusions

Efforts to make the ECU an international reserve asset and eventually a European common currency should be regarded as complementary. Market convertibility of the official ECU requires deep and resilient private markets. Interchangeability between the two sectors could be achieved without institutional changes if central banks could exchange their official deposits with

commercial banks, and commercial banks could simultaneously redeposit them with the BIS. The BIS, which also acts as a clearinghouse for commercial banks, is in fact empowered to hold official-ECU balances.

It might be argued that the whole scheme, even if acceptable from a formal point of view, has no real future given the reservations of the German monetary authorities concerning the role and function of the ECU. In reply, it should be recognized that Germany's legal argument is losing weight. The greater the role of the ECU in private as well as official markets, the more difficult it becomes to maintain that the ECU is merely an indexation device. The legal concept of money has already changed and will continue to be adapted to economic changes. The facts that there is no ECU "currency," no central bank behind it, and no legal payment in ECUs do not appear to be sufficient reasons for denying that the ECU has already acquired a wholly different monetary character from its predecessor, the EUA, which really *was* only an indexation instrument.

Official ECUs do exist, and they can be used for official settlements between central banks. The EMCF plays the role of a central institution overseeing the creation and distribution of official ECUs. Banking and foreign-exchange payments in ECUs, not only between private operators but also between market participants and official monetary authorities, amount to billions each year. Fines levied by the EC Commission can be paid in ECUs.

If market convertibility of the official ECU were to develop along the lines suggested here, it would become impossible for Germany not to acknowledge the monetary and foreign-exchange character of the ECU. I have shown that, under present conditions of uneven success in securing domestic monetary stability, certain central banks, notably the Bundesbank, manifest a preference for what I call "asymmetric monetary-base interventions" in EC currencies. I have also shown that this is precisely the character of intramarginal ECU interventions in the market, as opposed to mandatory interventions at the margin that involve settlements in official ECUs or recourse to the VSTF mechanism. The Bundesbank's attitude may not remain as negative as it is now. It may be forced to change by the liberalization of capital movements in ERM countries, and, especially if the United Kingdom joins the ERM, by the large additional volume of intervention that would be required to operate the system.

The counterpart to an expanded use of the official ECU would be an increasing role for the ECU as a substitute for domestic-currency borrowing and lending, leading to its use as a parallel common currency. This process would have to be closely monitored in order to prevent problems of money and credit control. The introduction of reserve requirements or other control instruments for all deposits—regardless of denomination—should be considered, in order to avoid artificial incentives for residents to favor ECU deposits rather than domestic-currency deposits held with the home banking system. This could be done either on a country-by-country basis, or, preferably, through the EMCF, which would be entrusted with the task of overseeing the process of ECU expansion (Thygesen, 1980). Convergence in monetary-control techniques would thereby be stimulated. Looking beyond the short-term frictions conected with individual countries' monetary policies, such a development would improve prospects for European monetary integration.

The high substitutability that would develop under this scenario between the ECU and each currency in the EMS would increase capital mobility within the Community, largely because of the intrinsic properties of the ECU, but also as a result of the "involuntary" capital flows that are likely to be set in motion. There is a certain irony in the fact that the countries most active and vociferous at the official level in pressing for further development of the ECU are also those which still impose capital controls.

If the two-pronged approach to developing the ECU suggested here were to be followed, the ECU could become an instrument for fostering European monetary integration without at the same time threatening monetary stability and credit control. It cannot be overemphasized, however, that this will be so if, and only if, efforts to continue the development of the official ECU and advances made by the markets in using it are not seen by Community governments and citizens as substitutes for the domestic monetary adjustments needed to achieve a zone of monetary stability.

I shall not dwell on these points; Robert Triffin and I have recently expounded them in great detail (Masera and Triffin, 1984). Let me simply stress that domestic cost-formation processes and public-sector deficits must be the primary target of economic-policy actions leading to integration. It is no accident that countries characterized by very high public deficits and high public debt/GDP ratios have encountered difficulties with the process of monetary and exchange-rate stabilization in the EMS. It should also be selfevident that capital-market liberalization in the EC is a prerequisite for any meaningful attempt to achieve greater monetary convergence and integration.

I conclude, therefore, on a sober note. While the ECU could be a useful instrument in building a European monetary dimension, the fact remains that the objective of gradual integration leading to monetary unification could be achieved even without the ECU. Finally, if the development of the ECU becomes an alibi for not tackling fundamental economic imbalances that impede domestic monetary stability, this potentially useful instrument will turn out to be counterproductive.

#### References

- Adler, Michael, and Bernard Dumas, "International Portfolio Choice and Corporation Finance: A Survey," *Journal of Finance*, 38 (June 1983), pp. 925-984.
- Allen, Polly Reynolds, *Birth of a New Euro-Currency: The ECU*, Occasional Paper No. 20, New York, Group of Thirty, 1986.
- Aschheim, Joseph, and Yoon S. Park, Artificial Currency Units: The Formation of Functional Currency Areas, Essays in International Finance No. 114, Princeton, N.J., Princeton University, International Finance Section, April 1976.
- Bilson, John F.O., "The Choice of an Invoice Currency in International Transactions," in J. S. Bhandari and B. H. Putnam, eds., *Economic Interdependence and Flexible Exchange Rates*, Cambridge, Mass., MIT Press, 1983, pp. 384-401.
- BIS, Fifty-fifth Annual Report, Basel, 1985.
- BIS, "Role of the BIS in a Private ECU Clearing System," February 1986, processed.
- Carbonetti, Francesco, "ECU," in *Enciclopedia Treccani Giuridica*, Rome, Istituto Giovanni Treccani, 1987.
- Ciampi, Carlo A., "The ECU: Problems and Perspectives," *Thema*, No. 6, Supplement, "International Banking System and the Use of the ECU," Istituto Bancario San Paolo di Torino, 1981, pp. 14-23.
- Coats, Warren L., Jr., "The SDR as a Means of Payment," IMF Staff Papers, 29 (September 1982), pp. 422-436.
- Committee of Governors of the Central Banks of the Member States of the European Economic Community—European Monetary Cooperation Fund, *Texts Concerning the European Monetary System and Related Documents*, December 1985.
- European Community Monetary Committee, Interim Report on the European Monetary System, September 1978.
- ECU Newsletter, "European Firms and ECU Invoicing," Istituto Bancario San Paolo di Torino, Survey No. 10, January 1986, pp. 3-33.
- Hamaui, Rony, "The ECU and Efficient Portfolio Choice," *Ricerche* (January 1985), Banca Commerciale Italiana.
- Harlandt, Hans, "Die ECU zulassen!" Zeitschrift für das gesamte Kreditwesen (No. 3, 1986).
- Jorion, Philippe, "The ECU and Efficient Portfolio Choice," paper presented at Conference on the ECU Market: Current Developments and Future Prospects, New York University, Jan. 30-31, 1986.
- Kenen, Peter B., "Use of the SDR to Supplement or Substitute for Other Means of Finance," in George M. von Furstenberg, ed., International Money and Credit: The Policy Roles, Washington, IMF, 1983, pp. 327-360.
- International Monetary Fund, International Monetary Reform: Documents of the Committee of Twenty, Washington, 1974.

- Levich, Richard M., "Developing the ECU Markets: Perspectives on Financial Innovation," paper presented at Conference on the ECU Market: Current Developments and Future Prospects, New York University, Jan. 30-31, 1986.
- Ludlow, Peter, The Making of the European Monetary System, London, Butterworth Scientific, 1982.

Magnifico, Giovanni, European Monetary Unification, London, Macmillan, 1973.

Markowitz, Harry M., "Portfolio Selection," Journal of Finance, 7 (March 1952), pp. 77-91.

------, Portfolio Selection, New York, Wiley, 1959.

Masera, Rainer S., "Deposit Creation, Multiplication and the Euro-Dollar Market," in *A Debate on the Eurodollar Market*, Rome, Luigi Einaudi Institute for Monetary, Banking, and Finance Studies, Research Paper No. 11, 1972.

-----, L'unificazione monetaria e lo SME, Bologna, Il Mulino, 1980.

-----, "Determinants of the Growth of the Euro-Currency Markets," Recent Developments in the Economic Analysis of the Euro-Markets, Basel, BIS, September 1982, pp. 3-28.

- Masera, Rainer S., and Robert Triffin, "Introduction" to Rainer S. Masera and Robert Triffin, eds., Europe's Money: Problems of European Monetary Coordination and Integration, Oxford, England, Clarendon Press, 1984, pp. 1-10.
- Mayer, Helmut W., Interaction between the Euro-Currency Markets and the Exchange Markets, BIS Economic Papers, No. 15, May 1985.
- Micossi, Stefano, "The Intervention and Financing Mechanism of the EMS and the Role of the ECU," *BNL Quarterly Review*, 38 (December 1985), pp. 327-345.
- Modigliani, Franco, and Giampiero Capponi, La finanza italiana, No. 12 (December 1985), pp. 4-8.
- Ossola, Rinaldo, *Towards New Monetary Relationships*, Essays in International Finance No. 87, Princeton, N.J., Princeton University, International Finance Section, July 1971.
- Padoa-Schioppa, Tommaso, "The EMF: Topics for Discussion" (presented at the Second International Seminar on European Economic and Monetary Union, Geneva, International Center for Monetary and Banking Studies, Dec. 7-8, 1979), BNL Quarterly Review, 33 (September 1980), pp. 317-343.
- ------, "Lessons from the European Monetary System," Bank of Italy Economic Bulletin, No. 2 (February 1986), pp. 63-72.
- Poehl, Karl O., "Le système monétaire européen: consolidation et évolution," speech by the President of the Deutsche Bundesbank at the French Association of Banks, Paris, July 1985.
- Sarcinelli, Mario, "Lo SME e il sistema monetario internazionale: verso una maggiore stabilità," *Moneta e Credito*, 38 (December 1985), pp. 371-403.
- Solnik, Bruno H., European Capital Markets: Towards a General Theory of International Investment, Lexington, Mass., Lexington Books, 1973.

——, "An Equilibrium Model of the International Capital Market," *Journal of Economic Theory*, 8 (August 1974), pp. 500-524.

- Swoboda, Alexander K., Credit Creation in the Euromarket: Alternative Theories and Implications for Control, Occasional Paper No. 2, New York, Group of Thirty, 1980.
- Thygesen, Niels, "Concluding Remarks" (presented at the Second International Seminar on European Economic and Monetary Union, Geneva, International Center for Monetary and Banking Studies, Dec. 7-8, 1979), BNL Quarterly Review, 33 (September 1980), pp. 382-387.
- Triffin, Robert, "How to End the World 'Infession': Crisis Management or Fundamental Reforms?" in R. S. Masera and R. Triffin, eds., 1984, pp. 13-76.
- Whalig, Berthold, Rechtsfragen zur privaten Verwendung der ECU in der Bundesrepublik Deutschland, Wertpapier Mitteilungen, No. 35, August 1985.
- Williamson, John, "Comment on Kenen," in George M. von Furstenberg, ed., International Money and Credit: The Policy Roles, Washington, IMF, 1983, pp. 367-373.

## STATISTICAL APPENDIX

## **EFFICIENT PORTFOLIO CHOICE AND THE ECU**

The BIS and the IMF data banks provided the basis for the data. Periodaverage domestic money-market interest rates and period-average exchange rates were used for EC currencies, the U.S. dollar, and the yen. For the ECU and the SDR, interest rates were calculated as a weighted average of domestic money-market rates on component currencies.

Efficient corner portfolios were derived according to Markowitz's (1959) critical-line method, assuming nonnegative investments. Input data were the variance-covariance matrixes and mean yields observed in two periods (1973I-1985IV) and (1979II-1985IV). The former covers the entire period of floating exchange rates among major currencies, the latter the period during which the EMS has been in effect. The shares of corner portfolio invested in single currencies are reported together with portfolio returns, standard deviations, and reward-to-variability ratios (Tables A. 1a and A. 1b). For every currency taken in turn as the base currency, corner portfolios are ranked, in decreasing order, from maximum return to minimum risk (Tables A.2a and A.2b). The efficient portfolios frontiers for the Italian lira and the Deutschemark are shown in Figures A.1 and A.2. See the Table of Currency Symbols below for the definitions of symbols used in the tables.

| Country        | Name         | Symbol |
|----------------|--------------|--------|
| Australia      | dollar       | A\$    |
| Canada         | dollar       | Can\$  |
| France         | franc        | FF     |
| Germany        | Deutschemark | DM     |
| Italy          | lira         | Lit    |
| Japan          | yen          | ¥      |
| Netherlands    | guilder      | DGlr   |
| Switzerland    | franc        | SwF    |
| United Kingdom | pound        | £      |
| United States  | dollar       | U.S.\$ |

TABLE OF CURRENCY SYMBOLS



FIGURE A.1 EFFICIENT PORTFOLIOS FRONTIER FOR A WEAK-CURRENCY COUNTRY, 1979II-1985IV (basis currency: lira)

NOTE: The mean and the standard deviation of yields of money-market investments in each currency are reported on the vertical and horizontal axes respectively.

FIGURE A.2 Efficient Portfolios Frontier for a Strong-Currency Country, 1979II-1985IV (basis currency: DM)



NOTE: The mean and the standard deviation of yields of money-market investments in each currency are reported on the vertical and horizontal axes respectively.

| Basis    |        |       |              |               |               |               |       |       |       |
|----------|--------|-------|--------------|---------------|---------------|---------------|-------|-------|-------|
| Currency | U.S.\$ | ¥     | DM           | FF            | £             | DGlr          | SDR   | ECU   | Lit   |
|          |        |       | Y            | ELDS ON FORE  | IGN INVESTMEN | <b>NTS</b>    |       |       |       |
|          |        |       |              | (annualize    | ed percent)   |               |       |       |       |
| U.S.\$   | 8.50   | 7.80  | 8.43         | 7.15          | 6.84          | 7.00          | 8.01  | 7.33  | 6.74  |
| ¥        | 6.10   | 5.34  | 5.99         | 4.83          | 4.53          | 4.58          | 5.62  | 4.99  | 4.54  |
| DM       | 7.31   | 6.58  | 7.21         | 6.00          | 5.69          | 5.80          | 6.82  | 6.16  | 5.65  |
| FF       | 12.45  | 11.85 | 12.42        | 10.95         | 10.63         | 10.96         | 11.95 | 11.17 | 10.37 |
| £        | 12.68  | 12.08 | 12.65        | 11.16         | 10.85         | 11.19         | 12.17 | 11.38 | 10.57 |
| DGlr     | 8.13   | 7.43  | 8.05         | 6.79          | 6.49          | 6.63          | 7.65  | 6.97  | 6.41  |
| SDR      | 8.93   | 8.25  | 8.86         | 7.56          | 7.25          | 7.43          | 8.44  | 7.74  | 7.14  |
| ECU      | 11.44  | 10.81 | 11.40        | 9.97          | 9.66          | 9.95          | 10.94 | 10.18 | 9.44  |
| Lit      | 17.58  | 17.11 | 17.62        | 15.88         | 15.55         | 16.11         | 17.05 | 16.15 | 15.07 |
|          |        |       | YIELDS ON FO | REIGN INVESTM | IENTS: STANDA | RD DEVIATIONS | 3     |       |       |
|          |        |       |              | (in pe        | ercent)       |               |       |       |       |
| U.S.\$   | 2.55   | 23.43 | 25.12        | 22.31         | 23.55         | 22.57         | 9.51  | 19.43 | 18.19 |
| ¥        | 22.37  | 0.88  | 23.53        | 20.80         | 24.02         | 21.92         | 16.12 | 19.25 | 19.08 |
| DM       | 24.85  | 22.30 | 2.83         | 12.31         | 20.47         | 6.81          | 15.32 | 8.39  | 14.33 |
| FF       | 25.86  | 21.46 | 14.05        | 2.43          | 21.49         | 12.80         | 15.81 | 9.48  | 13.82 |
| £        | 25.02  | 25.50 | 23.84        | 20.74         | 2.19          | 20.67         | 17.26 | 16.21 | 19.40 |
| DGlr     | 24.37  | 22.73 | 6.14         | 10.90         | 19.75         | 2.31          | 14.59 | 6.36  | 12.44 |
| SDR      | 11.60  | 16.67 | 16.27        | 13.68         | 17.35         | 14.19         | 1.95  | 10.32 | 11.55 |
| ECU      | 22.32  | 20.26 | 10.29        | 8.85          | 16.57         | 7.17          | 11.79 | 1.85  | 9.63  |
| Lit      | 23.65  | 23.51 | 19.36        | 13.97         | 20.29         | 15.50         | 15.08 | 10.37 | 2.59  |

 TABLE A. la

 Reward-to-Variability Ratios: Main Currencies and Currency Baskets, 1973I-1985IV

|            | (in percent) |        |        |        |        |        |              |        |        |  |
|------------|--------------|--------|--------|--------|--------|--------|--------------|--------|--------|--|
| U.S.\$     | 333.50       | 33.31  | 33.54  | 32.03  | 29.03  | 31.00  | 84.22        | 37.71  | 37.08  |  |
| 关          | 27.26        | 608.10 | 25.46  | 23.25  | 18.86  | 20.91  | 34.84        | 25.92  | 23.79  |  |
| DM         | 29.39        | 29.50  | 254.60 | 48.71  | 27.79  | 85.05  | <b>44.51</b> | 73.45  | 39.40  |  |
| <b>F</b> F | 48.15        | 55.25  | 88.45  | 450.00 | 49.47  | 85.65  | 75.57        | 117.80 | 75.01  |  |
| £          | 50.66        | 47.38  | 53.06  | 53.83  | 495.10 | 54.12  | 70.51        | 70.23  | 54.49  |  |
| DGlr       | 33.37        | 32.67  | 131.20 | 62.34  | 32.83  | 287.30 | 52.41        | 109.60 | 51.49  |  |
| SDR        | 76.99        | 49.45  | 54.47  | 55.28  | 41.80  | 52.34  | 432.30       | 75.06  | 61.82  |  |
| ECU        | 51.25        | 53.39  | 110.80 | 112.70 | 58.27  | 138.80 | 92.78        | 550.60 | 97.99  |  |
| Lit        | 74.34        | 72.78  | 90.99  | 113.70 | 76.66  | 104.00 | 113.10       | 155.70 | 582.90 |  |

#### FOREIGN INVESTMENTS: REWARD-TO-VARIABILITY RATIOS

| Basis    |        |       |              |                            |                              |               |       |       |       |
|----------|--------|-------|--------------|----------------------------|------------------------------|---------------|-------|-------|-------|
| Currency | U.S.\$ | ¥     | DM           | FF                         | £                            | DGlr          | SDR   | ECU   | Lit   |
|          |        |       | Y            | ields on fore<br>(annualiz | IGN INVESTMEN<br>ed percent) | VTS           |       |       |       |
| U.S.\$   | 10.48  | 6.14  | 3.35         | 3.98                       | 6.32                         | 2.80          | 7.44  | 4.59  | 5.96  |
| ¥        | 9.72   | 5.37  | 2.63         | 3.29                       | 5.60                         | 2.08          | 7.00  | 3.88  | 5.28  |
| DM       | 15.36  | 11.05 | 8.00         | 8.43                       | 10.93                        | 7.42          | 12.20 | 9.14  | 10.32 |
| FF       | 19.96  | 15.69 | 12.39        | 12.63                      | 15.27                        | 11.79         | 16.70 | 13.43 | 14.44 |
| £        | 16.35  | 12.06 | 8.95         | 9.34                       | 11.87                        | 8.37          | 13.18 | 10.07 | 11.21 |
| DGlr     | 15.89  | 11.59 | 8.51         | 8.92                       | 11.43                        | 7.93          | 12.73 | 9.64  | 10.80 |
| SDR      | 12.86  | 8.53  | 5.61         | 6.15                       | 8.56                         | 5.05          | 9.76  | 6.81  | 8.09  |
| ECU      | 17.62  | 13.33 | 10.15        | 10.50                      | 13.06                        | 9.57          | 14.41 | 11.25 | 12.34 |
| Lit      | 22.29  | 18.04 | 14.61        | 14.77                      | 17.48                        | 14.00         | 18.98 | 15.61 | 16.52 |
|          |        |       | YIELDS ON FO | REIGN INVESTM              | IENTS: STANDA                | RD DEVIATIONS | 3     |       |       |
|          |        |       |              | (in pe                     | ercent)                      |               |       |       |       |
| U.S.\$   | 2.10   | 24.94 | 23.36        | 23.87                      | 27.78                        | 24.61         | 10.17 | 22.00 | 19.65 |
| ¥        | 23.55  | 0.55  | 21.40        | 20.30                      | 25.11                        | 21.65         | 16.62 | 19.57 | 17.73 |
| DM       | 25.41  | 22.83 | 2.25         | 6.58                       | 21.01                        | 4.45          | 14.63 | 4.45  | 6.67  |
| FF       | 27.98  | 22.39 | 9.87         | 1.90                       | 22.58                        | 10.01         | 16.67 | 7.51  | 9.71  |
| £        | 28.58  | 27.68 | 20.96        | 21.28                      | 2.05                         | 20.07         | 19.47 | 17.18 | 20.78 |
| DGlr     | 26.51  | 23.34 | 3.61         | 6.41                       | 20.81                        | 2.03          | 15.37 | 4.31  | 6.97  |
| SDR      | 12.22  | 17.68 | 13.78        | 13.91                      | 20.21                        | 14.93         | 1.70  | 11.72 | 9.97  |
| ECU      | 24.95  | 21.32 | 6.04         | 5.49                       | 18.09                        | 6.24          | 13.26 | 1.73  | 6.24  |
| Lit      | 24.95  | 21.86 | 8.67         | 7.00                       | 21.34                        | 9.40          | 13.05 | 5.98  | 2.36  |

 TABLE A.1b

 Reward-to-Variability Ratios: Main Currencies and Currency Baskets, 1979II-1985IV

|        | (in percent) |        |        |        |        |        |        |        |        |  |  |  |  |
|--------|--------------|--------|--------|--------|--------|--------|--------|--------|--------|--|--|--|--|
| U.S.\$ | 498.40       | 24.62  | 14.35  | 16.67  | 22.76  | 11.38  | 73.22  | 20.86  | 30.34  |  |  |  |  |
| ¥      | 41.29        | 971.90 | 12.27  | 16.18  | 22.31  | 9.60   | 40.31  | 19.82  | 29.79  |  |  |  |  |
| ĎМ     | 60.44        | 48.41  | 356.10 | 128.10 | 52.01  | 166.80 | 83.40  | 205.40 | 154.80 |  |  |  |  |
| FF     | 71.34        | 70.06  | 125.50 | 664.30 | 67.64  | 117.80 | 100.10 | 179.00 | 148.80 |  |  |  |  |
| £      | 57.22        | 43.56  | 42.69  | 43.91  | 578.80 | 41.70  | 67.69  | 58.61  | 53.97  |  |  |  |  |
| ĎĜlr   | 59.97        | 49.68  | 236.10 | 139.10 | 54.94  | 390.60 | 82.83  | 223.80 | 154.90 |  |  |  |  |
| SDR    | 105.20       | 48.26  | 40.75  | 44.21  | 42.38  | 33.83  | 573.40 | 58.06  | 81.05  |  |  |  |  |
| ECU    | 70.62        | 62.52  | 168.00 | 191.20 | 72.18  | 153.30 | 108.70 | 650.10 | 197.70 |  |  |  |  |
| Lit    | 89.36        | 82.52  | 168.60 | 211.10 | 81.89  | 148.90 | 145.40 | 261.20 | 700.70 |  |  |  |  |

1

#### FOREIGN INVESTMENTS: REWARD-TO-VARIABILITY RATIOS

| U.S.\$ | ¥      | DM     | FF     | £ | DGlr        | SDR         | ECU    | Lit    | E     | S     | E/S(%)     |
|--------|--------|--------|--------|---|-------------|-------------|--------|--------|-------|-------|------------|
|        |        |        |        |   | BASIS CURRE | NCY: U.S.\$ |        |        |       |       |            |
| _      | 0      | 1      | 0      | 0 | 0           | 0           | 0      | 0      | 8.43  | 25.12 | 33.54      |
|        | 0      | 0      | 0      | 0 | 0           | 1           | 0      | 0      | 8.01  | 9.51  | 84.22 (*)  |
|        |        |        |        |   | BASIS CUR   | RENCY: ¥    |        |        |       |       |            |
| 1      | _      | 0      | 0      | 0 | 0           | 0           | 0      | 0      | 6.10  | 22.37 | 27.26      |
| 0.5657 | _      | 0.4343 | 0      | 0 | 0           | 0           | 0      | 0      | 6.05  | 19.10 | 31.69      |
| 0      | _      | 0.1243 | 0      | 0 | 0           | 0.8757      | 0      | 0      | 5.66  | 16.29 | 34.77      |
| 0      |        | 0.0048 | 0      | 0 | 0           | 0.9952      | 0      | 0      | 5.62  | 16.12 | 34.85      |
| 0      | _      | 0      | 0.0074 | 0 | 0           | 0.9926      | 0      | 0      | 5.61  | 16.12 | 34.81      |
| 0      | —      | 0      | 0.0455 | 0 | 0           | 0.9545      | 0      | 0      | 5.58  | 16.10 | 34.66      |
| 0      | _      | 0      | 0.0467 | 0 | 0           | 0.9184      | 0      | 0.0349 | 5.54  | 16.10 | 34.43 (*)  |
|        |        |        |        |   | BASIS CURR  | ency: DM    |        |        |       |       |            |
| 1      | 0      | _      | 0      | 0 | 0           | 0           | 0      | 0      | 7.31  | 24.85 | 29.39      |
| 0      | 0      | _      | 0      | 0 | 0           | 1           | 0      | 0      | 6.82  | 15.32 | 44.51      |
| 0      | 0      |        | 0      | 0 | 0           | 0.3998      | 0.6002 | 0      | 6.43  | 10.42 | 61.69      |
| 0      | 0      |        | 0      | 0 | 0.2966      | 0.2294      | 0.4739 | 0      | 6.21  | 8.36  | 74.25      |
| 0      | 0.0545 |        | 0      | 0 | 0.6486      | 0           | 0.2969 | 0      | 5.95  | 6.81  | 87.33      |
| 0      | 0.0543 | —      | 0      | 0 | 0.7966      | 0           | 0.1491 | 0      | 5.89  | 6.68  | 88.25      |
| 0      | 0.0541 | —      | 0.0343 | 0 | 0.8950      | 0           | 0.0166 | 0      | 5.85  | 6.65  | 87.99 (*)  |
|        |        |        |        |   | BASIS CURR  | ENCY: FF    |        |        |       |       |            |
| 1      | 0      | 0      | _      | 0 | 0           | 0           | 0      | 0      | 12.45 | 25.86 | 48.15      |
| 0.1687 | 0      | 0.8313 | _      | 0 | 0           | 0           | 0      | 0      | 12.43 | 13.39 | 92.84      |
| 0.0836 | 0.1608 | 0.7556 | _      | 0 | 0           | 0           | 0      | 0      | 12.34 | 12.48 | 98.85      |
| 0      | 0.1394 | 0.7055 | _      | 0 | 0           | 0.1551      | 0      | 0      | 12.27 | 12.13 | 101.10     |
| 0      | 0.1439 | 0.6785 | _      | 0 | 0           | 0.1776      | 0      | 0      | 12.26 | 12.07 | 101.50     |
| 0      | 0.1302 | 0.3517 |        | 0 | 0           | 0           | 0.5181 | 0      | 11.70 | 10.18 | 114.90     |
| 0      | 0.0554 | 0      | —      | 0 | 0           | 0           | 0.9446 | 0      | 11.20 | 9.42  | 118.90     |
| 0      | 0.0508 | 0      |        | 0 | 0           | 0           | 0.9492 | 0      | 11.20 | 9.42  | 118.90 (*) |

 TABLE A.2a

 Composition of Efficient Portfolios, 1973I-1985IV

.

|         |  |        |        |        | BASIS CUI  | RRENCY: £  |        |        |       |       |            |
|---------|--|--------|--------|--------|------------|--|--------|--------|-------|-------|------------|
| 1       | 0                                      | 0      | 0      |        | 0          | 0  | 0      | 0      | 12.68 | 25.02 | 50.66      |
| 0.4687  | 0                                      | 0.5313 | 0      |        | 0          | 0  | 0      | 0      | 12.66 | 20.13 | 62.90      |
| 0.4402  | 0.0528                                 | 0.5070 | 0      |        | 0          | 0  | 0      | 0      | 12.63 | 19.87 | 63.58      |
| 0.2230  | 0                                      | 0.3904 | 0      |        | 0          | 0.3866   | 0      | 0      | 12.47 | 18.67 | 66.80      |
| 0       | 0                                      | 0.2610 | 0      |        | 0          | 0.7390   | 0      | 0      | 12.29 | 17.44 | 70.48      |
| 0       | 0                                      | 0.1864 | 0      |        | Ó          | 0.8136   | 0      | 0      | 12.26 | 17.26 | 71.03      |
| 0       | 0                                      | 0      | 0      |        | 0          | 0.6630   | 0.3370 | 0      | 11.90 | 16.06 | 74.14      |
| 0       | 0                                      | 0      | 0      |        | 0          | 0.3606   | 0.6394 | 0      | 11.67 | 15.69 | 74.34 (*)  |
| <u></u> |  |        |        | Ë      | BASIS CURP | ENCY: DGlr   |        |        |       |       |            |
| 1       | 0                                      | 0      | 0      | 0      |            | 0  | 0      | 0      | 8.13  | 24.37 | 33.37      |
| 0.0565  | 0                                      | 0.9435 | 0      | Ó      |            | 0  | 0      | 0      | 8.06  | 6.00  | 134.40     |
| 0.0556  | 0.0016                                 | 0.9428 | Ó      | 0      | -          | Ö  | 0      | 0      | 8.06  | 5.99  | 134.50     |
| 0.0483  | 0                                      | 0.9138 | 0      | 0      | C10004     | 0  | 0.0379 | 0      | 8.02  | 5.79  | 138.60     |
| Ô       | 0                                      | 0.7357 | 0      | 0      | -          | 0  | 0.2643 | 0      | 7.77  | 4.69  | 165.50     |
| 0       | 0                                      | 0.5165 | 0      | 0      | <u></u>    | 0  | 0.4835 | 0      | 7.53  | 4.24  | 177.60 (*) |
|         | ······································ |        |        | 1      | BASIS CURE | rency: SDR   |        |        |       |       |            |
| 1       | 0                                      | 0      | 0      | 0      | 0          |  | 0      | 0      | 8.93  | 11.60 | 76.99      |
| 0.6145  | 0                                      | 0.3855 | 0      | 0      | 0          | -  | Ô      | 0      | 8.90  | 5.62  | 158.30     |
| 0.5241  | 0.1546                                 | 0.3213 | 0      | 0      | 0          | mand link.   | 0      | 0      | 8.80  | 3.97  | 222.00     |
| 0.5046  | 0.1606                                 | 0.3007 | 0      | 0.0341 | 0          |  | Ô      | Ô      | 8.74  | 3.55  | 246.10     |
| 0.4609  | 0.1497                                 | 0.1956 | 0.1123 | 0.0816 | 0          |  | 0      | 0      | 8.52  | 2.55  | 334.00     |
| 0.4376  | 0.1439                                 | 0.1224 | 0.1135 | 0.0766 | 0          | and the second s | 0.1060 | 0      | 8.41  | 2.40  | 350.00     |
| 0.4346  | 0.1430                                 | 0.1160 | 0.1142 | 0.0769 | 0          |  | 0.1114 | 0.0039 | 8.40  | 2.40  | 349.70 (*) |

| U.S.\$ | ¥      | DM     | FF     | £      | DGlr        | SDR       | ECU    | Lit      | E     | S     | E/S(%)     |
|--------|--------|--------|--------|--------|-------------|-----------|--------|----------|-------|-------|------------|
|        |        |        |        |        | BASIS CURRI | ENCY: ECU |        |          |       |       |            |
| 1      | 0      | 0      | 0      | 0      | 0           | 0         |        | 0        | 11.44 | 22.32 | 51.25      |
| 0.2175 | 0      | 0.7825 | 0      | 0      | 0           | 0         |        | 0        | 11.41 | 8.70  | 131.10     |
| 0.1744 | 0.0806 | 0.7450 | 0      | 0      | 0           | 0         |        | 0        | 11.36 | 8.22  | 138.20     |
| 0      | 0.0385 | 0.6409 | 0      | 0      | 0           | 0.3206    |        | 0        | 11.23 | 7.35  | 152.80     |
| 0      | 0.0391 | 0.6386 | 0      | 0      | 0           | 0.3224    | _      | 0        | 11.23 | 7.34  | 152.90     |
| 0      | 0.0403 | 0.6313 | 0      | 0.0142 | 0           | 0.3142    | _      | 0        | 11.20 | 7.21  | 155.40     |
| 0      | 0.0347 | 0.5634 | 0.0828 | 0.0514 | 0           | 0.2678    | _      | 0        | 11.05 | 6.33  | 174.70     |
| 0      | 0.0224 | 0.4080 | 0.1852 | 0.1193 | 0           | 0.0791    |        | 0.1860   | 10.51 | 3.99  | 263.70     |
| 0      | 0.0290 | 0.1903 | 0.2090 | 0.1338 | 0.2241      | 0         |        | 0.2138   | 10.11 | 3.17  | 319.30     |
| 0      | 0.0252 | 0.1250 | 0.2153 | 0.1332 | 0.2875      | 0         | _      | 0.2138   | 10.01 | 3.13  | 320.30 (*) |
|        |        |        |        |        | BASIS CURF  | ENCY: Lit |        |          |       |       |            |
| 0      | 0      | 1      | 0      | 0      | 0           | 0         | 0      |          | 17.62 | 19.36 | 90.99      |
| 0.3751 | 0      | 0.6249 | 0      | 0      | 0           | 0         | 0      |          | 17.60 | 16.12 | 109.20     |
| 0.3265 | 0.1014 | 0.5721 | 0      | 0      | 0           | 0         | 0      | <u> </u> | 17.55 | 15.72 | 111.70     |
| 0.1598 | 0.0379 | 0      | 0      | 0      | 0           | 0         | 0.8023 |          | 16.41 | 10.53 | 155.80     |
| 0.1319 | 0.0240 | 0      | 0      | 0      | 0           | 0         | 0.8441 |          | 16.36 | 10.37 | 157.70     |
| 0.0954 | 0      | 0      | 0.0427 | 0      | 0           | 0         | 0.8618 |          | 16.27 | 10.23 | 159.00     |
| 0.0753 | 0      | 0      | 0.0574 | 0      | 0           | 0         | 0.8673 |          | 16.24 | 10.22 | 158.90 (*) |

TABLE A.2a (cont.)Composition of Efficient Portfolios, 1973I-1985IV

....

E = Portfolio return.

S = Portfolio standard deviation.

(\*) = Minimum risk portfolio.

| U.S.\$ | ¥              | ĎМ | FF     | £      | DĞlr        | ŜĎŔ         | ĔĊU    | Ĺit    | Е     | S     | E/S(%)     |
|--------|----------------|----|--------|--------|-------------|-------------|--------|--------|-------|-------|------------|
|        |                |    |        | 1      | BASIS CURRE | NGY: U.S.\$ |        |        |       |       |            |
|        | 0              | 0  | 0      | 0      | 0           | 1           | 0      | 0      | 7.44  | 10.17 | 73.22      |
|        |                |    |        |        | BASIS CUR   | RENCY: ¥    |        |        |       |       |            |
| 1      |                | 0  | 0      | 0      | .0          | 0           | 0      | 0      | 9.72  | 23.55 | 41.29      |
| 0.8051 | al or defined. | 0  | 0      | 0.1949 | 0           | 0           | 0      | 0      | 8.92  | 20.96 | 42.56      |
| 0.6537 |                | 0  | 0      | 0.1664 | 0           | 0           | 0      | 0.1799 | 8.24  | 19.42 | 42.42      |
| 0.3452 |                | 0  | 0      | 0.1005 | 0           | 0.5543      | 0      | 0      | 7.63  | 18.23 | 41.87      |
| 0      | 000000         | 0  | 0      | 0.0041 | 0           | 0.9959      | 0      | 0      | 6.69  | 16.61 | 40.30      |
| 0      |                | 0  | 0      | 0.0254 | 0           | 0.9746      | 0      | 0      | 6.67  | 16.58 | 40.23      |
| 0      | <u></u>        | 0  | 0      | 0.0234 | 0           | 0.7055      | 0      | 0.2711 | 6.29  | 16.38 | 38.39 (*)  |
|        |                |    |        |        | BASIS CURRI | ency: DM    |        |        |       |       |            |
| 1      | 0              | -  | 0      | 0      | 0           | 0           | 0      | 0      | 15.36 | 25.41 | 60.44      |
| 0.9225 | 0              |    | 0      | 0.0775 | 0           | 0           | 0      | 0      | 15.01 | 23.89 | 62.85      |
| 0.1221 | 0              |    | 0      | 0.0567 | 0           | 0           | 0      | 0.8212 | 10.97 | 8.07  | 135.90     |
| 0.0798 | 0              |    | 0      | 0      | 0           | 0           | 0.3037 | 0.6165 | 10.36 | 6.37  | 162.70     |
| 0      | 0              |    | 0      | 0      | 0           | 0           | 0.5643 | 0.4357 | 9.65  | 4.71  | 205.10     |
| 0      | 0              |    | 0      | 0      | 0           | 0           | 0.6072 | 0.3928 | 9.60  | 4.62  | 207.80     |
| Ö      | 0              |    | 0      | 0      | 0.2837      | 0           | 0.4388 | 0.2775 | 8.98  | 3.83  | 234.30     |
| 0      | 0              |    | 0.0784 | 0      | 0.5147      | 0           | 0.2374 | 0.1695 | 8.40  | 3.57  | 235.20 (*) |
|        |                |    |        |        | BASIS CURR  | ency: FF    |        |        |       |       |            |
| 1      | 0              | 0  |        | 0      | 0           | 0           | 0      | 0      | 19.96 | 27.98 | 71.34      |
| 0.8922 | 0              | 0  |        | 0.1078 | 0           | 0           | 0      | 0      | 19.45 | 25.88 | 75.15      |
| 0.8463 | 0              | 0  |        | 0.1055 | Ô           | 0           | 0      | 0.0481 | 19.20 | 24.89 | 77.14      |
| 0.2228 | 0.0505         | 0  |        | 0.0661 | 0           | 0           | Ô      | 0.6605 | 15.79 | 12.65 | 124.70     |
| 0.1404 | 0.0624         | 0  |        | 0      | 0           | 0           | 0.3367 | 0.4605 | 14.95 | 10.28 | 145.50     |
| 0      | 0.0653         | 0  |        | 0      | 0           | 0           | 0.6438 | 0.2909 | 13.87 | 7.83  | 177.10     |
| 0      | 0.0268         | 0  |        | 0      | 0           | 0           | 0.9732 | 0      | 13.49 | 7.49  | 180.20     |
| 0      | 0.0233         | 0  |        | 0      | 0           | 0           | 0.9767 | 0      | 13.49 | 7.49  | 180.10 (*) |

TABLE A.2bComposition of Efficient Portfolios, 1979II-1985IV

| U.S.\$ | ¥      | DM     | FF     | £      | DGlr        | SDR        | ECU    | Lit    | E     | s     | E/S(%)     |
|--------|--------|--------|--------|--------|-------------|------------|--------|--------|-------|-------|------------|
|        |        |        |        |        | BASIS CUR   | RENCY: £   |        |        |       |       |            |
| 1      | 0      | 0      | 0      | _      | 0           | 0          | 0      | 0      | 16.35 | 28.58 | 57.22      |
| 0.9322 | 0.0677 | 0      | 0      |        | 0           | 0          | 0      | 0      | 16.06 | 27.60 | 58.20      |
| 0.5917 | 0      | 0      | 0      | _      | 0           | 0.4083     | 0      | 0      | 15.06 | 24.48 | 61.50      |
| 0      | 0      | 0      | 0      | _      | 0           | 1          | 0      | 0      | 13.18 | 19.47 | 67.69      |
| 0      | 0      | 0      | 0      | _      | 0           | 0.2253     | 0.7747 | 0      | 10.77 | 16.95 | 63.53 (*)  |
|        |        |        |        | I      | BASIS CURRI | ENCY: DGlr |        |        |       |       |            |
| 1      | 0      | 0      | 0      | 0      | _           | 0          | 0      | 0      | 15.89 | 26.51 | 59.97      |
| 0.9023 | 0      | 0      | 0      | 0.0977 | _           | 0          | 0      | 0      | 15.46 | 24.53 | 63.01      |
| 0.0896 | 0      | 0      | 0      | 0.0821 | —           | 0          | 0      | 0.8283 | 11.31 | 8.03  | 140.80     |
| 0.0277 | 0      | 0      | 0      | 0      | _           | 0          | 0.4452 | 0.5271 | 10.43 | 5.53  | 188.50     |
| 0      | 0      | 0      | 0      | 0      |             | 0          | 0.5380 | 0.4620 | 10.18 | 4.89  | 208.00     |
| 0      | 0      | 0      | 0      | 0      |             | . 0        | 0.5510 | 0.4490 | 10.16 | 4.86  | 209.20     |
| 0      | 0      | 0.3457 | 0      | 0      |             | 0          | 0.4548 | 0.1995 | 9.48  | 3.52  | 269.60     |
| 0      | 0      | 0.5948 | 0.1251 | 0      |             | 0          | 0.2802 | 0      | 8.88  | 3.01  | 294.70     |
| 0      | 0      | 0.6140 | 0.1334 | 0      | —           | 0          | 0.2526 | 0      | 8.85  | 3.01  | 293.90 (*) |
|        |        |        |        | ]      | BASIS CURR  | ency: SDR  |        |        |       |       |            |
| 1      | 0      | 0      | 0      | 0      | 0           |            | 0      | 0      | 12.86 | 12.22 | 105.20     |
| 0.9050 | 0      | 0      | 0      | 0.0950 | 0           | _          | 0      | 0      | 12.45 | 10.21 | 122.00     |
| 0.8372 | 0.0425 | 0      | 0      | 0.1202 | 0           |            | 0      | 0      | 12.16 | 8.92  | 136.30     |
| 0.4695 | 0.0862 | 0      | 0      | 0.1064 | 0           | -          | 0      | 0.3378 | 10.42 | 2.35  | 443.30     |
| 0.4391 | 0.0902 | 0      | 0.1114 | 0.0978 | 0           | _          | 0      | 0.2616 | 10.05 | 1.94  | 518.80     |
| 0.4366 | 0.0911 | 0      | 0.1118 | 0.0945 | 0           | _          | 0.0158 | 0.2501 | 10.02 | 1.93  | 517.70 (*) |

TABLE A.2b (cont.)Composition of Efficient Portfolios, 1979II-1985IV

• 1

|        |        |        |        | B      | ASIS CURF | iency: ECU |           |            |       |       |            |
|--------|--------|--------|--------|--------|-----------|------------|-----------|------------|-------|-------|------------|
| 1      | Ŏ      | 0      | Ô      | 0      | 0         | 0          | -         | 0          | 17.62 | 24.95 | 70.62      |
| 0.7849 | 0      | 0      | 0      | 0.2151 | 0         | 0          |           | 0          | 16.64 | 20.38 | 81.64      |
| 0.1419 | 0      | 0      | 0      | 0.1916 | 0         | 0          | -         | 0.6665     | 13.23 | 6.96  | 190.00     |
| 0.0421 | 0      | 0      | 0.3193 | 0.1736 | 0         | 0          | ments     | 0.4650     | 12.10 | 3.73  | 324.70     |
| 0      | 0      | 0.1815 | 0.3798 | 0.1594 | 0         | 0          | grant the | 0.2792     | 11.36 | 2.39  | 476.00     |
| 0      | 0      | 0.2585 | 0.4201 | 0.1510 | 0         | 0          | _         | 0.1704     | 11.11 | 2.25  | 493.10 (*) |
|        |        |        |        |        | BASIS CUP | RENCY: Lit |           |            |       |       |            |
| 1      | 0      | 0      | 0      | 0      | 0         | 0          | 0         |            | 22.29 | 24.95 | 89.36      |
| 0.8152 | 0      | 0      | 0      | 0.1848 | 0         | 0          | 0         | unimation. | 21.40 | 21.16 | 101.20     |
| 0.7556 | 0.0335 | Ó      | 0      | 0.2109 | 0         | 0          | 0         |            | 21.13 | 20.17 | 104.80     |
| 0.3771 | 0.0432 | Û.     | 0      | 0.0381 | 0         | 0          | 0.5416    |            | 18.31 | 10.52 | 174.00     |
| 0.3152 | 0.0416 | 0      | 0.0558 | 0.0182 | 0         | 0          | 0.5692    |            | 17.81 | 8.97  | 198.40     |
| 0.1792 | 0.0176 | 0      | 0.0891 | 0      | 0         | 0.2183     | 0.4958    | -          | 17.51 | 8.11  | 215.80     |
| 0.0885 | 0      | 0      | 0.1418 | 0      | 0         | 0.3539     | 0.4157    |            | 17.27 | 7.45  | 231.70     |
| 0      | 0      | 0      | 0.2278 | 0      | 0         | 0.4137     | 0.3585    |            | 16.81 | 6.31  | 266.60     |
| 0      | 0      | 0      | 0.4180 | 0      | 0         | 0.1776     | 0.4044    |            | 15.86 | 5.14  | 308.30 (*) |

E = Portfolio return.

S = Portfolio standard deviation. (\*) = Minimum risk portfolio.

# PUBLICATIONS OF THE INTERNATIONAL FINANCE SECTION

### Notice to Contributors

The International Finance Section publishes at irregular intervals papers in four series: ESSAYS IN INTERNATIONAL FINANCE, PRINCETON STUDIES IN INTERNA-TIONAL FINANCE, SPECIAL PAPERS IN INTERNATIONAL ECONOMICS, and REPRINTS IN INTERNATIONAL FINANCE. ESSAYS and STUDIES are confined to subjects in international finance. SPECIAL PAPERS are surveys of the literature suitable for courses in colleges and universities. The Section welcomes the submission of manuscripts for these series.

An ESSAY should be a lucid exposition of a theme, accessible not only to the professional economist but to other interested readers. It should therefore try to avoid technical terms, eschew mathematics and statistical tables (except when essential for an understanding of the text), and use footnotes sparingly.

A STUDY OF SPECIAL PAPER may be more technical. It may include statistics and algebra and may have many footnotes. STUDIES and SPECIAL PAPERS may also be longer than ESSAYS; indeed, these two series are meant to accommodate manuscripts too long for journal articles and too short for books.

To facilitate prompt evaluation, please submit three copies of your manuscript. Retain one for your files. The manuscript should be typed on one side of 8½ by 11 strong white paper. All material should be double-spaced—text, excerpts, footnotes, tables, references, and figure legends. For more complete guidance, prospective contributors should send for the Section's style guide before preparing their manuscripts.

### How to Obtain Publications

A mailing list is maintained for free distribution of all new publications to college, university, and public libraries and nongovernmental, nonprofit research institutions.

Individuals and organizations not qualifying for free distribution can obtain ESSAYS and REPRINTS as issued and announcements of new STUDIES and SPECIAL PAPERS by paying a fee of \$12 (within U.S.) or \$15 (outside U.S.) to cover the period January 1 through December 31, 1987. Alternatively, for \$30 they can receive all publications automatically—SPECIAL PAPERS and STUDIES as well as ESSAYS and REPRINTS.

ESSAYS and REPRINTS can also be ordered from the Section at \$4.50 per copy, and STUDIES and SPECIAL PAPERS at \$6.50. Payment MUST be included with the order and MUST be made in U.S. dollars. PLEASE INCLUDE \$1 FOR POSTACE AND HANDLING. (These charges are waived on orders from persons or organizations in countries whose foreign-exchange regulations prohibit such remittances.) For airmail delivery outside U.S., Canada, and Mexico, there is an additional charge of \$1.50.

All manuscripts, correspondence, and orders should be addressed to:

International Finance Section Department of Economics, Dickinson Hall

Princeton University Princeton, New Jersey 08544

Subscribers should notify the Section promptly of a change of address, giving the old address as well as the new one.

## List of Recent Publications

Some earlier issues are still in print. Write the Section for information.

## ESSAYS IN INTERNATIONAL FINANCE

- 140. Pieter Korteweg, Exchange-Rate Policy, Monetary Policy, and Real Exchange-Rate Variability. (Dec. 1980)
- 141. Bela Balassa, The Process of Industrial Development and Alternative Development Strategies. (Dec. 1980)
- 142. Benjamin J. Cohen, The European Monetary System: An Outsider's View. (June 1981)
- 143. Marina v. N. Whitman, International Trade and Investment: Two Perspectives. (July 1981)
- 144. Sidney Dell, On Being Grandmotherly: The Evolution of IMF Conditionality. (Oct. 1981)
- 145. Ronald I. McKinnon and Donald J. Mathieson, How to Manage a Repressed Economy. (Dec. 1981)
- \*146. Bahram Nowzad, The IMF and Its Critics. (Dec. 1981)
- 147. Edmar Lisboa Bacha and Carlos F. Díaz Alejandro, International Financial Intermediation: A Long and Tropical View. (May 1982)
- 148. Alan A. Rabin and Leland B. Yeager, Monetary Approaches to the Balance of Payments and Exchange Rates. (Nov. 1982)
- 149. C. Fred Bergsten, Rudiger Dornbusch, Jacob A. Frenkel, Steven W. Kohlhagen, Luigi Spaventa, and Thomas D. Willett, From Rambouillet to Versailles: A Symposium. (Dec. 1982)
- 150. Robert E. Baldwin, The Inefficacy of Trade Policy. (Dec. 1982)
- 151. Jack Guttentag and Richard Herring, The Lender-of-Last Resort Function in an International Context. (May 1983)
- 152. G. K. Helleiner, The IMF and Africa in the 1980s. (July 1983)
- 153. Rachel McCulloch, Unexpected Real Consequences of Floating Exchange Rates. (Aug. 1983)
- 154. Robert M. Dunn, Jr., The Many Disappointments of Floating Exchange Rates. (Dec. 1983)
- 155. Stephen Marris, Managing the World Economy: Will We Ever Learn? (Oct. 1984)
- 156. Sebastian Edwards, The Order of Liberalization of the External Sector in Developing Countries. (Dec. 1984)
- 157. Wilfred J. Ethier and Richard C. Marston, eds., with Kindleberger, Guttentag and Herring, Wallich, Henderson, and Hinshaw, International Financial Markets and Capital Movements: A Symposium in Honor of Arthur I. Bloomfield. (Sept. 1985)
- 158. Charles E. Dumas, The Effects of Government Deficits: A Comparative Analysis of Crowding Out. (Oct. 1985)
- Jeffrey A. Frankel, Six Possible Meanings of "Overvaluation": The 1981-85 Dollar. (Dec. 1985)

\* Out of print. Available on demand in xerographic paperback or library-bound copies from University Microfilms International, Box 1467, Ann Arbor, Michigan 48106, United States, and 30-32 Mortimer St., London, WIN 7RA, England. Paperback reprints are usually \$20. Microfilm of all Essays by year is also available from University Microfilms. Photocopied sheets of out-ofprint titles are available on demand from the Section at \$6 per Essay and \$8 per Study or Special Paper.

- Stanley W. Black, Learning from Adversity: Policy Responses to Two Oil Shocks. (Dec. 1985)
- 161. Alexis Rieffel, The Role of the Paris Club in Managing Debt Problems. (Dec. 1985)
- 162. Stephen E. Haynes, Michael M. Hutchison, and Raymond F. Mikesell, Japanese Financial Policies and the U.S. Trade Deficit. (April 1986)
- 163. Arminio Fraga, German Reparations and Brazilian Debt: A Comparative Study. (July 1986)
- 164. Jack M. Guttentag and Richard J. Herring, Disaster Myopia in International Banking. (Sept. 1986)
- 165. Rudiger Dornbusch, Inflation, Exchange Rates, and Stabilization. (Oct. 1986)
- 166. John Spraos, IMF Conditionality: Ineffectual, Inefficient, Mistargeted. (Dec. 1986)
- 167. Rainer Stefano Masera, An Increasing Role for the ECU: A Character in Search of a Script. (June 1987)

## PRINCETON STUDIES IN INTERNATIONAL FINANCE

- 45. Ian M. Drummond, London, Washington, and the Management of the Franc, 1936-39. (Nov. 1979)
- 46. Susan Howson, Sterling's Managed Float: The Operations of the Exchange Equalisation Account, 1932-39. (Nov. 1980)
- 47. Jonathan Eaton and Mark Gersovitz, Poor Country Borrowing in Private Financial Markets and the Repudiation Issue. (June 1981)
- 48. Barry J. Eichengreen, Sterling and the Tariff, 1929-32. (Sept. 1981)
- 49. Peter Bernholz, Flexible Exchange Rates in Historical Perspective. (July 1982)
- 50. Victor Argy, Exchange-Rate Management in Theory and Practice. (Oct. 1982)
- Paul Wonnacott, U.S. Intervention in the Exchange Market for DM, 1977-80. (Dec. 1982)
- 52. Irving B. Kravis and Robert E. Lipsey, Toward an Explanation of National Price Levels. (Nov. 1983)
- 53. Avraham Ben-Bassat, Reserve-Currency Diversification and the Substitution Account. (March 1984)
- \*54. Jeffrey Sachs, Theoretical Issues in International Borrowing. (July 1984)
- 55. Marsha R. Shelburn, Rules for Regulating Intervention under a Managed Float. (Dec. 1984)
- Paul De Grauwe, Marc Janssens, and Hilde Leliaert, Real-Exchange-Rate Variability from 1920 to 1926 and 1973 to 1982. (Sept. 1985)
- 57. Stephen S. Golub, The Current-Account Balance and the Dollar: 1977-78 and 1983-84. (Oct. 1986)
- 58. John T. Cuddington, Capital Flight: Estimates, Issues, and Explanations. (Dec. 1986)
- 59. Vincent P. Crawford, International Lending, Long-Term Credit Relationships, and Dynamic Contract Theory. (March 1987)

## SPECIAL PAPERS IN INTERNATIONAL ECONOMICS

- 10. Richard E. Caves, International Trade, International Investment, and Imperfect Markets. (Nov. 1974)
- \*11. Edward Tower and Thomas D. Willett, The Theory of Optimum Currency Areas and Exchange-Rate Flexibility. (May 1976)

- \*12. Ronald W. Jones, "Two-ness" in Trade Theory: Costs and Benefits. (April 1977)
- 13. Louka T. Katseli-Papaefstratiou, The Reemergence of the Purchasing Power Parity Doctrine in the 1970s. (Dec. 1979)
- \*14. Morris Goldstein, Have Flexible Exchange Rates Handicapped Macroeconomic Policy? (June 1980)
- 15. Gene M. Grossman and J. David Richardson, Strategic Trade Policy: A Survey of Issues and Early Analysis. (April 1985)

#### REPRINTS IN INTERNATIONAL FINANCE

- William H. Branson, Asset Markets and Relative Prices in Exchange Rate Determination. [Reprinted from Sozialwissenschaftliche Annalen, Vol. 1, 1977.] (June 1980)
- 21. Peter B. Kenen, The Analytics of a Substitution Account. [Reprinted from Banca Nazionale del Lavoro Quarterly Review, No. 139 (Dec. 1981).] (Dec. 1981)
- Jorge Braga de Macedo, Exchange Rate Behavior with Currency Inconvertibility. [Reprinted from Journal of International Economics, 12 (Feb. 1982).] (Sept. 1982)
- Peter B. Kenen, Use of the SDR to Supplement or Substitute for Other Means of Finance. [Reprinted from George M. von Furstenberg, ed., International Money and Credit: The Policy Roles, Washington, IMF, 1983, Chap. 7.] (Dec. 1983)
- 24. Peter B. Kenen, Forward Rates, Interest Rates, and Expectations under Alternative Exchange Rate Regimes. [Reprinted from Economic Record, 61 (Sept. 1985).] (June 1986)
- 25. Jorge Braga de Macedo, Trade and Financial Interdependence under Flexible Exchange Rates: The Pacific Area. [Reprinted from Augustine H.H. Tan and Basant Kapur, eds., Pacific Growth and Financial Interdependence, Sydney, Australia, Allen and Unwin, 1986, Chap. 13.] (June 1986)

.

ISBN 0-88165-074-9